INTERNATIONAL STUDENTS' JOB SEARCH STRATEGIES IN THE OIL AND GAS INDUSTRY OF NEWFOUNDLAND AND LABRADOR

by

© Sepideh Gorjizadeh

A Thesis Submitted to the

School of Graduate Studies

in partial fulfillment of the requirements for the degree of

Master's of Education

Faculty of Education

Memorial Univerity of Newfoundland

August 2015

St. John's

Newfoundland and Labrador

ABSTRACT

In light of the periodic skill shortage problems in the oil and gas industry in Newfoundland and Labrador, this study uncovered the job search strategies utilized by international graduate students in the oil and gas engineering and earth sciences programs enrolled at Memorial University of Newfoundland and Labrador. The study also identified factors influencing students' choices of job search strategies and investigated the role of the Memorial University, as a higher education institution in the Province, in providing services and support in students' career search process. The researcher used semi-structured interviews to collect data. The findings indicated that students used networking and online job searches the most to seek employment, followed by the career services offered by the University Career Centre (UCC). Students chose search strategies based on several factors: the type of job sought; the influence of peers and word of mouth; the industry of interest; previous work experience; and the role of the UCC, which is considered to be both a search strategy and influencing factor. Two themes emerged from the data analysis: a) job search as a planned project and b) students' perceptions of the oil and gas labour market of the Province.

ACKNOWLEDGEMENTS

I wholeheartedly thank my mother, who has given me unlimited support during my master's studies at Memorial University. I am grateful to my supervisor Dr. Dale Kirby and Associate Dean of the Education Faculty of Memorial University, Dr. Tim Seifert, for their guidance and support during my thesis writing process. Also, I would like to express my gratitude to Nancy Bishop, Graduate Programs Secretary, for making efforts in providing me with necessary logistical resources, and Jenna C. Flogeras for providing insightful comments on my thesis. Special thanks go to the staff of the Career Development and Experiential Learning Department of Memorial University of Newfoundland for giving me valuable information and background knowledge about their career programs and services to students, which have greatly informed my study.

TABLE OF CONTENTS

ABSTRACTii
ACKNOWLEDGEMENTS
LIST OF TABLES
LIST OF ABBREVIATIONS ix
LIST OF APPENDICES x
CHAPTER 1. INTRODUCTION
CHAPTER 2. LITERATURE
2.1 Introduction
2.2 Newfoundland's Oil and Gas Labour Market
2.3 Job Search Behaviour in Newfoundland and Labrador
2.4 Higher Education
2.5 Job Search
2.6 Job Search Strategies
2.6.1 Formal and informal strategies
2.6.2 Networking
2.6.3 Weak ties and strong ties
2.7 Other Job Search Strategies
2.7.1 Internet job search
2.7.1.1 LinkedIn
2.7.2 University career centres

2.8 Factors Influencing Choice of Job Search Strategies	
2.8.1 Primary factors	
2.8.1.1 Race	
2.8.1.2 Age	
2.8.1.3 Gender	27
2.8.1.4 Education	
2.8.1.5 Social status.	
2.8.2 Secondary factors	
2.8.2.1 Information richness.	
2.8.2.2 Cost, convenience, and usefulness	
2.8.2.3 Personality traits	
2. 9 Conclusion	
CHAPTER 3. METHODOLOGY	
3.1 Introduction	
3.2 Research Design	
3.3 Data Collection and Sampling	
3.4 Data Analysis	
3.5 Ethical Considerations	40
3.6 Summary	
CHAPTER 4. FINDINGS AND ANALYSIS	
4.1 Introduction	
4.2 Job Search Strategies	
4.2.1 Networking.	

4.2.2 Online job search	52
4.2.2.1 LinkedIn	
4.2.2.2 Job search websites.	55
4.2.2.3 Applying directly through the company website.	57
4.2.2.4 Professional associations.	58
4.3 Influencing Factors on Choosing Job Search Strategies	61
4.3.1 Type of job	61
4.3.2 Peers and word-of-mouth.	61
4.3.3 Type of industry	65
4.3.4 Having prior work experience.	67
4.3.5 University Career Centre (UCC).	69
4.4 Emergent Themes	71
4.4.1 Job searching as a planned project.	71
4.4.2 Students' perceptions of the oil and gas labour market of the province	77
4.4.3 Recommendations from the students	
CHAPTER 5. DISCUSSION	
5.1 Introduction	
5.2 International Students' Job Search Process	83
5.3 Students' Perceptions of the Oil and Gas Labour Market of the Province	100
CHAPTER 6. RECOMMENDATIONS AND CONCLUSIONS	105
6.1 Recommendations from the Researcher	105
6.2 Conclusion	106
REFERENCES	109

BIBLIOGRAPHY	
APPENDICES	
Appendix A	
Appendix B	
Appendix C	
Appendix D	
Appendix E	
Appendix F	

LIST OF TABLES

Title of table	Page number
Table 1. Graduate Student Enrolment at Memorial University by program and orig	gin-Fall
2014	13

LIST OF ABBREVIATIONS

APA	American Psychological Association
ETP	Entrepreneurship Training Program
HR	Human Resources
ICEHR	Interdisciplinary Committee on Ethics in Human Research
IPA	Interpretive Phenomenological Analysis
MUN	Memorial University of Newfoundland and Labrador
NL	Newfoundland and labrador
NOIA	Newfoundland and Labrador Oil and Gas Industries Association
PES	Public employment services
R&D	Research and Development
PR	Permanent Residence
PSDP	Professional Skills Development Program
UCC	University Career Centre

LIST OF APPENDICES

Title of Append	ix	Page Number
Appendix A	Informed consent form for students	
Appendix B	Informed consent form for administration	
Appendix C	Recruitment script for students	
Appendix D	Recruitment script for administration	
Appendix E	Interview questionnaire for students	141
Appendix F	Interview questionnaire for administration	143

Chapter 1

Introduction

This chapter presents the problem under study, research aims and questions, and introduces the subsequent chapters along with their objectives. This study is aimed at minimizing the gap in the knowledge of job search strategies used by international postgraduate students enrolled in Oil and Gas Engineering and Earth Sciences programs in Newfoundland and Labrador by understanding the factors affecting their choice of strategies and identifying the role of higher education institutions in the province in their job search process. The participants for this study were selected among the international graduate students at Memorial University of Newfoundland who were in their final year of studies or their third year of doctoral studies.

This topic of inquiry was chosen for research in light of the periodic skill shortage problems in the oil and gas industry in Newfoundland and Labrador (Newfoundland and Labrador, 2011). The Petroleum Human Resources Council of Canada (2013) has predicted there will be between 37,700 and 47,900 job openings in this sector within the next ten years. As a result, it has been suggested that "the inability to adequately source the staffing requirements could jeopardize successful project execution" (Petro Canada, 2007, p. 73) and "there is a risk that the companies in this sector may have difficulty sourcing the labour needed for current and future operations" (Suncor Energy Inc., 2012, p. 71).

This problem is highlighted by the fact that a skilled workforce already exists in the Province, which includes international students studying in oil and gas industry-related disciplines (with or without prior industry experience). Consequently, outsourcing temporary foreign workers was proposed by the Department of Advanced Education and Skills of the

Newfoundland and Labrador (NL) Government (2007) as a solution to the perceived problem of skills shortage in this industry.

The purpose of this study is to a) provide an understanding of job search strategies that international graduate students in the Oil and Gas Engineering and Earth Sciences programs when they search for employment within Newfoundland and Labrador; b) identify factors that influence students' choices of such job search strategies; and c) investigate the role of provincial post-secondary education institutions in providing services and support during the students' career search process.

The individual research questions were as follows:

- What job search strategies do international graduate students in the Oil and Gas Engineering and Earth Sciences programs in Newfoundland and Labrador use to look for jobs in the province's oil and gas industry?
- 2. What are the factors influencing oil and gas student job seekers' choices of job search strategies?
- 3. What supports, guidance, and resources do higher education institutions in Newfoundland and Labrador provide to students who are seeking employment opportunities in the oil and gas industry?

This study did not intend to provide successful job search strategies; rather, it attempted to find out what strategies the students were using, the strategies they prefered to use, and why they used them. This study was intended to provide valuable information relevant to those providing job search and career advice at post-secondary institutions in the province. The findings may also provide industry policy makers with a better understanding of the barriers in our knowledge of

the existing, potential workforce and employment policies within the Newfoundland and Labrador oil and gas industry.

In addition, this research report explored the factors that influenced students' job search choices and strategies within Newfoundland and Labrador, and took into consideration the students' perceptions, attitudes, and feelings about their employment opportunities and prospects in the context of the local labour market in the Newfoundland and Labrador oil and gas industry.

Theories of job search deal with unemployed search models and study a wide range of job seeker characteristics as they relate to the labour market, from skills match to wage distribution, quality of jobs, job satisfaction, job vacancies, unemployment duration, search intensity, and search strategies (Devine & Kiefer, 1993). These theories focus on job search characteristics of the unemployed, which could include active search strategies because the unemployed person is more likely to search actively for a job than an employed person who mainly browses his/her options for a job change (Hempel, 2010). Job search strategy as one of the characteristics of jobseekers, whether employed or unemployed, also applies to the participants of this study because some international graduate students had been previously employed while others had no prior work experience.

This introductory chapter is followed by five subsequent chapters. Chapter 2 provides a background to the main issues related to student job search in the oil and gas industry in Newfoundland and Labrador by presenting studies through literature review. The literature review focuses on previous research studies that have included graduate students or recent post-secondary graduates. Chapter 2 presents a collection of studies that contain arguments regarding the influencing factors on the students' choices of job search strategies.

Chapter 3 provides a description of the methodology used in this study. In this chapter, the research design is presented: a phenomenological, qualitative approach to the research questions was adopted and semi-structured interviews were selected as a means of data collection. A description of how the data were collected is included along with a rationale for the process used for selecting participants. This chapter also outlines how the data were analyzed. Also, this study sought to cover all ethical guidelines.

Chapter 4 presents the findings of the study with a focus on pre-theme categorization based on the research questions. For the analysis, qualitative data was coded, categorized, and themed and detailed descriptions of the findings were presented. Interpretive Phenomenological Analysis (IPA) was chosen an additional method of data analysis to analyze the feelings, perceptions, and attitudes of the students toward the local labour market.

In Chapter 5 the findings are discussed in the context of the relevant literature described in Chapter 2.

Finally, Chapter 6 presents recommendations and conclusions of this study. Implications presented in this chapter are established through the lens of students' experiences, followed by the researcher's recommendations addressing policy makers and career centres. The knowledge developed through this research is expected to have in the areas of career program planning by career centres and employment policy making by the government.

Chapter 2

Literature

2.1 Introduction

This chapter first reviews Newfoundland and Labrador's labour market as it pertains to the oil and gas industry in particular. The following section is a comprehensive overview of previous research on job searches, job search strategies, and job search behaviour in Newfoundland and Labrador, specifically. Job search strategies of interest to this study include formal and informal strategies, networking, "weak ties" and "strong ties" in one's social capital¹, the use of online job search strategies, and utilization of the University Career Centre (UCC) at Memorial University. This section also includes some of the factors that influence choosing between possible job search strategies and/or resources.

More focus is given to those studies related to post-secondary student and graduate populations; however, the overall unemployed worker population cannot be ignored, since some of the study participants are also experienced professionals who were taking a break from fulltime employment and intending to return to the labour market after graduation. The broad literature on job search strategies identifies a variety of factors that have been found to influence individual strategies. As such, the influencing factors are categorized into two groups. "Primary factors" are more prominent in the literature and include race, age, gender, education, and social status. Those have been termed "secondary factors" were not the primary focus in the arguments of the studies, such as information richness, cost, convenience and usefulness, and personality traits.

¹ Social Capital is defined as informal or formal contacts in one's network. (Ben-Porath, 1980)

There are a variety of human and social factors that affect job search choices: self-esteem (Ellis & Taylor, 1983); career objectives (Boswell, Boudreau, & Dunford, 2004; Van Hoye & Saks, 2008); corporate and recruitment image (Gatewood, Gowan, & Lautenschlager, 1993); language fluency, social support, and cultural knowledge (Guerrero & Rothstein, 2012); and location of the job seeker (e.g., urban vs. rural) (Weber & Mahringer, 2012). A particularly notable observation is that students and other jobseekers who present to have low self-esteem choose strategies that tend to involve less social networking compared to those with high self-esteem (Ellis & Taylor, 1983). Nevertheless, some authors such as Thomas (1989) contend that self-esteem does not significantly influence one's search strategy choices.

Also, access to information about jobs such as vacancies and the quality of jobs can affect the choice of job search strategy (Ford & Purdue University, 1974); however, no specific strategies are mentioned. On the other hand, access to information about job vacancies is mediated by company-independent sources, such as word of mouth, and company-dependent sources (e.g., the company's recruitment activities) such as advertisements in the media (Van Hoye, Klehe, & Van Hooft, 2014).

There have been numerous studies of job search strategies including quantitative surveys, qualitative interview-based research, and case studies. Studies show that among both postsecondary student and graduate populations, some job search techniques are used more than others, but on the whole, they are generally aware of the strategies available to them (McKeown & Lindorff, 2011). Moreover, the choice of strategies is affected by many of the factors mentioned above. As a result, it is important to understand what job search strategies students and graduates choose and why they choose them because, eventually, that is what bridges the gap between one's knowledge of job vacancies and how they are filled.

According to the literature that was reviewed for this research, the job search strategies most commonly used by students and graduates are informal search methods such as establishing social contacts, either through social networking or using social media for online networking, engaging the services provided at post-secondary institutions, and contacting employers directly (Huffman & Torres, 2001; Johnson, 1997; Kim, 1994; Try, 2005). Students and graduate jobseekers viewed these methods as the most effective approaches. On the other hand, studies of unemployed jobseekers show that they mostly prefer to use social networking and newspaper ads as informal and formal methods, respectively. Also, the unemployed jobseekers' low socio-economic background might lead them to use public or private employment agencies (OECD, 1999; Try, 2005).

2.2 Newfoundland's Oil and Gas Labour Market

The production of oil and gas in 12 of the 13 Canadian provinces and territories makes the oil and gas industry "the largest single private sector investor" in the entire country (Canadian Association of Petroleum Producers [CAPP], 2014, para. 1). Statistics Canada (2012) categorizes the oil and gas industry as *Oil & Gas Extraction*, which includes those industries with activities ranging from operation and extraction to production and shipment of oil, gas, and hydrocarbon liquids. In 2013, the number of Newfoundland and Labrador businesses in this industry was estimated at 91 (Department of Finance of the NL Government, March 2014), with 50 of these located in the St. John's metropolitan area (Department of Finance of the NL Government, March 2014). The only oil refinery, transshipment terminal, dry-docks, and fabrication facilities in the province are located in the Clarenville-Bonavista region. This includes the dry-dock and fabrication facility in Bull Arm, the transshipment terminal for offshore oil in Whiffen Head, and the long-standing refinery in Come By Chance (Skeard et al., 2011).

With four producing oil projects and two additional projects under development (Newfoundland and Labrador Oil and Gas Industries, n.d), it is projected that most new job growth in the Province will occur in the oil and gas industry (Department of Human Resources of the NL Government, 2011). The mining and oil sector produced 367,400 jobs in Canada in 2013 (Department of Finance of the NL Government, March 2014) and represents 4.4% of employment share (Department of Human Resources of the NL Government, 2011) with 16,000 employees in Newfoundland and Labrador (Department of Finance of the NL Government, April 2015). The oil and gas sector is regarded by the provincial government as one of the few sectors that will grow faster than average over the entire forecast period.

The provincial employment projections of this industry predict the need for a skilled workforce and raise the question of how the skilled workers are sourced in this industry. Although higher education institutes in the province offer oil and gas-related programs of study, the construction and extraction jobs in this industry need little formal education and offer on-thejob training to employees (Lockar & Wolf, 2012). On the other hand, there are jobs that require formal education, such as reservoir engineer or geologist positions. At the same time, "there is limited information about [employment success of] graduates from Newfoundland and Labrador education institutions" (Petroleum Industry Human Resources Committee, 2001, p. 15) in the oil and gas industry of the province.

With respect to this argument, labour availability is the most important business issue human resources (HR) professionals are faced with (Ernst & Young, 2011). Petroleum Human Resources Council of Canada (2013) has reported that nearly 90% of industry employers regard employee recruitment as their company's primary mechanism for dealing with workforce issues, while the most significant skills shortages are associated with project/field engineers and geoscientists (Ernst & Young, 2011).

2.3 Job Search Behaviour in Newfoundland and Labrador

Osberg (1993) used a fishing analogy for job search behaviour. That is, during the job search process, jobseekers use different methods to catch the "big fish." Sometimes they throw fish back into the water, as the kind of job they are looking for might be specific. Therefore, knowing the intentions of student jobseekers is important to their search process (Boswell et al., 2004). The fishing analogy can apply to companies and recruitment agencies as well because they try to attract the best talent; on the other hand, universities and post-secondary institutions aim to prepare the students with the skills required for their future career by providing career transition programs such as internships, co-operative education, and apprenticeships for the students.

In a 1993 study, Seyfrit found that students in the province of Newfoundland and Labrador lacked awareness of oil and gas projects and prospective job opportunities arising in the sector. The study also found that when they did enroll in oil and gas-related post-secondary programs, older students tended to enroll in technical/vocational programs whereas younger students were more likely to attend university (Seyfrit, 1993). Today, oil and gas-related careers in the Province are among the 50 occupational groups that are projected to undergo the strongest job growth and employee recruitment pressures over the 2011 to 2020 period. This includes positions in the "upstream" oil and gas industry such as oil and gas drillers, operators, and supervisors (Department of Human Resources of the NL Government, 2011). This means it will be essential for students to be aware of the potential career opportunities in the Province's oil and

gas industry. Hence, the job search behaviour of students in the post-secondary schools of the Province is of great importance.

While some students may choose oil and gas-related fields as their future occupation, many will also choose to work elsewhere. This is especially the case for Newfoundlanders and Labradoreans, who have demonstrated high degrees of labour mobility (Hill, 1983) as workers seek out employment opportunities both within and outside the Province. In 2010, for example, this mobile workforce included approximately 76% of mobile workers who left the Province to work in other areas of Canada, primarily Alberta. This percentage represented somewhere between 7,890 and 10,600 individuals from the Province's overall available workforce (Department of Human Resources of the NL Government, 2011, p. 45). Over the past 30 years, there has been a noticeable increase in the Province's workforce mobility, which has coincided with the growth in oil sands production in Alberta. Despite moving in search of better work prospects, many of these Newfoundlanders and Labradoreans frequently relocate back to their home Province (Singh & Devine, 2013), as they regard their temporary outmigration to be only a "short-term economic strategy" (Hill, 1983, p. 228). This fits with one of the mandates of the Newfoundland and Labrador Skills Task Force initiative: to recruit and retain skilled workers who have left the Province (Department of Advanced Education and Skills of the NL Government, 2007). This initiative also focused on skilled foreign nationals who come to the Province for study or work and have frequently been hired in managerial and professional jobs (Hill, 1983).

The Report of the Skills Task Force (Department of Advanced Education and Skills of the NL Government, 2007) provided a variety of recommendations to the K-12 school system, post-secondary institutions, government, and employers. Some of these recommendations

involve government funding initiatives geared toward labour market factors such as student and graduate employment, employee retention, skills development, industry partnership, and networking. Among the numerous policy recommendations of the Skills Task Force, there is no reference to the existing workforce of international students in the Province who had prior work experience before coming to study in Canada. Despite this, in many instances international students face workforce barriers comparable to those experienced by foreign workers, including a lack of recognition for their skills, experience, and educational credentials (Department of Advanced Education and Skills of the NL Government, 2007).

2.4 Higher Education

Memorial University of Newfoundland (MUN) is the only university in the Province. Several departments offer oil and gas-related programs at MUN: the Faculty of Engineering, with a degree in Oil and Gas Engineering and related degrees such as Mechanical and Electrical Engineering; Applied Sciences, including the Department of Earth Sciences, with degrees in Geoscience, Geophysics, and Geochemistry; and the Marine Institute (Oil & Gas Development Partnership, 2013). College of the North Atlantic and other private institutes are also higher education options. At College of the North Atlantic, students can study in a three-year diploma program in Petroleum Engineering Technology and Process Operations Engineering Technology.

Memorial University and private institutes offer co-operative education, internships, and apprenticeships to students in undergraduate and diploma programs (College of the North Atlantic, n.d). The Globe and Mail (2008) coined Memorial co-op studies as a "try before you buy" opportunity, with more than 1000 students in its co-op engineering programs. Co-operative education, internships, and apprenticeships are a means of workplace learning that focus on

content, as well as context, with an experiential learning approach towards adult education (O'Connor, 2004); however, these are not mandatory components of graduate programs.

On the other hand, career preparation courses can increase students' career adaptability and help them in their career explorations (Koen et al., 2012). Career resources at MUN are offered at the career centre. University-industry partnership is another opportunity for students to gain exposure to the industry. With the Province producing 40% of Canada's oil, Memorial University was the only university in the North Atlantic region to initiate the Oil and Gas Development Partnership (OGDP) in the year 2000 (Oil & Gas Development Partnership, 2013). This partnership benefits students in graduate courses with a focus on research. The Research and Development Corporation (RDC) invested 2.3 million dollars in Memorial-led research and development (R&D) projects on the St. John's campus in July 2014 (Today.mun.ca). Universityindustry partnerships provide more research opportunities for both the faculty members and students at Memorial University. Also, students can take advantage of expanding their professional networks, acquire information about the oil and gas companies and their projects in the Province, and explore career options.

Table 1 provides a breakdown of graduate student enrollment in Engineering and Applied Sciences programs at Memorial University of Newfoundland during the Fall 2014 semester. While specific data are not available, a sub-population of these international students are enrolled in oil and gas engineering programs and earth sciences programs. Table 1

Graduate Student Enrolment at Memorial University by program and origin-Fall 2014²

Origin		Program
Newfoundland and Labrador	1,715 ³	Engineering and Applied Sciences 5594
Out-of-province	844 ⁴	
International	1,0383	
Graduate Total	3,579	

2.5 Job Search

Research on approaches to job searching has focused on a wide range of different populations, from unemployed youth to undergraduate and graduate students and alumni looking for jobs. Much of the literature research on job search strategies has been conducted using quantitative approaches that employ surveys.

One of the most influential and relevant studies of job search strategies is Try's (2005) examination of strategies used by graduates. This study, carried out among Norwegian students and graduates between 1995 and 2000, administered a survey to investigate the use of successful job search methods. Approximately 30% of the survey population was comprised of graduates.

 $^{^2}$ The focus of this study is on the graduate student population at Memorial University, hence the table illustrates graduate student enrolment for Fall semester 2014, including full-time and part-time, and students from Newfoundland, out-of-province, and international locations.

³ Memorial University of Newfoundland - Fact Book 2014, Table 13, Out-of-province Full-time and Part-time Undergraduate and Graduate Enrolment - Fall Semester. Retrieved from http://www.mun.ca/ciap/Analysis/Factbook/Factbook2014.php

⁴ Memorial University of Newfoundland - Fact Book 2014, Table 5A, Undergraduate and Graduate Enrolment by Faculty/School - Fall Semester. Retrieved from http://www.mun.ca/ciap/Analysis/Factbook/Factbook2014.php

The results of this study indicated that the most used method of job search was approaching employers directly by following up on their job advertisements. The second most popular method was accessing the public employment office.

Another study by Royal (2001) examined the job search process used by 176 undergraduate students employed in part-time jobs. Results showed that students favored four types of job search methods: personal contacts, job advertisements, university-related methods such as the university career centre and on-campus job postings, and direct applications to prospective employers. The findings of this study are consistent with those of Marmaros and Sacerdote (2002), who found that graduating students mainly turned to their immediate social contacts (i.e., family, friends, and teachers) in seeking potential job opportunities. University career search services were also among their main job search strategies.

Other studies have focused on job search strategies of young "Gen Y" alumni: people born between 1982 and 2003 (Howe & Strauss, 2000). Bissola and Imperatori (2013) carried out one such qualitative study designed to provide recommendations on social media job search strategies of Gen Yers for employers and recruiters. They mainly questioned the use of social media as a job search tool by young, employed alumni in the Italian labour market. This study found that participants used company websites and a university career service as initial key tools in looking for a job. Also, while a variety of social media were used by jobseekers, LinkedIn was the predominant one. This study is significant in its focus to bridge the gap between alumni job search practices and the strategies used by recruiters. Bissola and Imperatori recognized the importance of the internet and online presence for the young generation of jobseekers. They emphasized that employers must pay more attention to their online branding with this in mind.

Online tools were also one of the main focuses of McKeown and Lindorff (2011), who interviewed 45 recent Australian university graduates. They found that all the graduates used the internet to conduct online job searches. Furthermore, most graduates were unaware of the services available from the university career centre. New graduates also had high expectations of their potential for future employment until trial and error showed they had to lower their expectations and accept less lucrative jobs in order to enter the labour market. McKeown and Lindorff also presented a variety of recommendations for university career centres that are intended to help them provide students with programs that can instill a more realistic sense of labour market opportunities. The need to increase awareness of university career services is echoed in a study by Gerard (2011), who pointed out that although on-campus career centres may provide an array of career-related services, students are frequently unfamiliar with the services that exist beyond traditional services such as resume and interview workshops.

Other recent studies of student and recent graduate job search strategies have shown that these populations favor using their personal networks, university career services, and online job search tools. However, Monahan, Shah, and Jochum (2014) concluded that one of the main tools utilized by college students in their job search was newspaper advertisements. This traditional job search activity (as noted by Kimmell, Keltyka, Olsen, & Kaplan, 2003) is considered to be passive if jobseekers only look through the advertisements for job postings, and it becomes an active job search strategy once they actually apply to jobs after reading the advertisements (Royal, 2001).

Some of the research studies in the literature examined distinct populations such as students in specific majors (Kimmell et al., 2003; Ng & Burke, 2006; Rate & Boost, 2012). Ng and Burke (2006) contextualized and compared the experiences of undergraduate cooperative

and non-cooperative business students in a comparative analysis. The researchers found that the students who completed a cooperative component as part of their studies had a better understanding of the job market, employer needs, and future jobs. Consequently, these students were more confident in carrying out their job search process compared to the students who did not have this cooperative education experience.

Kimmell et al. (2003) and Rate and Boost (2012) investigated the job search choices of graduate students in the area of accounting and business. Both works highlighted the wide variety of job search strategies utilized by graduate students in these two disciplines. Kimmell, et al. (2003) found that the accounting students in particular relied on a wider variety of resources to look for a job. The most commonly used methods were the university career centre, business contacts, job fairs, temporary employment agencies, general and discipline-specific job search websites, newspaper advertisements, and company websites. The researchers argued that although students did not have a positive opinion of career centre services, the majority of students initially used these university job search services first. They suggested, given the importance of student access to employers, campus career resources should place a greater emphasis on employer outreach and ensure they have a more active presence on campus in order to reach students.

Research conducted by Rate and Boost (2012) supports these findings. The study participants, who were MBA graduates, regarded their personal contacts and networks to be the most effective resource in job hunting. Consequently, they used their personal network of contacts to successfully locate job openings. It is also noteworthy that they too took advantage of university career centre job search services.

The research literature pertaining to job search strategies has not focused exclusively on post-secondary student and graduate populations. For example, a report on the job search strategies of unemployed jobseekers in the U.S. from 1970 to 1971 stated that the three most-used methods were a) sending applications directly to the employers, b) utilizing the services of public and private job agencies, and c) posting personalized "job-wanted" advertisements (Bradshaw, 1973). A later study of unemployed youth in the U.S., carried out by Holzer (1986), also indicated a preference for using one's close contacts such as friends and relatives, in addition to sending applications directly to prospective employers. More recently, a similar study showed that jobseekers were most likely to use media advertisements and personal contacts in their job search first, followed consecutively by approaching public employment agencies and sending applications directly to employers (Weber & Mahringer, 2008).

2.6 Job Search Strategies

Job search behaviour consists of the formal and informal strategies used by a jobseeker in order to look for a job (Royal, 2001; Van Hoye et al., 2014). This includes all the methods that jobseekers might choose as part of their job search behaviour; however, more research is required to uncover the specific sources used in job searches (Van Hoye et al., 2014). Thomas (1989) divided job search behaviours into three categories: traditional, non-traditional, and networking. In his study of business, science, and engineering students, he categorized sending resumes to employers and calling or visiting employers as non-traditional search strategies. He classified responding to newspaper advertisements, employer interviews at placement agencies, and seeking the services of off-campus employment agencies under the scope of traditional strategies. The third category, networking, included contacting faculty, academic advisors, alumni, family, friends, and/or placement services staff.

Thomas, Crossley and Highhouse (2005) divided job search strategies into three quite different categories: focused, exploratory, and haphazard. People who adopt a focused search strategy narrow down and concentrate on their career objectives, while exploratory searching involves examining and sampling different career and search options. Jobseekers using a haphazard search strategy consider a wide range of career options and do not limit themselves to specific employment option. Royal (2001) divided job search strategies into *active* and *passive*. Passive job search methods include seeking out and reading newspaper ads, while following through and applying for the job vacancies found through these ads is active behaviour. Royal (2001) also considered browsing the internet and job agencies online to be more in the realm of passive behaviour compared to actually sending out resumes and responding to online job postings.

2.6.1 Formal and informal strategies. The research literature on job search strategies distinguishes between formal and informal job search strategies. The former include searching through newspaper advertisements, engaging with employment agencies (McKenna, 1985; Royal 2001; Try, 2005), using advertisements of public or private employment services (Try, 2005; Van Hoye 2009), and accessing campus recruitment services (Van Hoye 2009). Informal strategies include learning about job opportunities by word of mouth (McKenna, 1985), indirect approaches to potential employers through relatives (Royal, 2001; Van Hoye, 2009), social networks (Try, 2005), friends or acquaintances (Try, 2005; Royal, 2001; Van Hoye, 2009), teachers, current or former employees, approaching a previous employer for re-hire, and submitting resumes in person to the companies (Van Hoye, 2009). It is also argued that informal job searching can provide jobseekers with more accurate and detailed information about job vacancies, which may lead to more successful job interviews (Van Hoye, 2013).

Previous research has examined a variety digital job search strategies: utilizing specialized online services (Bissola & Imperatori, 2013) and online job boards (e.g., Monster.com; Hempel, 2010; Nikolaou, 2014), company recruitment websites (Bissola & Imperatori, 2013), university graduate placement and career services (Bissola & Imperatori, 2013; Gerard, 2011; Royal, 2001), online campus job postings (Royal, 2001), social networking websites (Herbold & Douma, 2013; Nikolaou, 2014; Scheele & Madison, 2010; Sin & Kim, 2013) such as LinkedIn (Adams, 2013; Garge & Telang, 2011; Hempel, 2010; Sin & Kim, 2013; Sułkowski & Chmielecki, 2014), and receiving emailed job alerts from companies (McKeown & Lindorff, 2011).

Research has also investigated the role of experiential and on-the-job training opportunities, such as co-operative education placements and internships for post-secondary students, in facilitating successful student transitions to the labour market. Although these opportunities to gain work experience and build relationships with potential employers cannot be referred to as search tools or strategies, they can be valuable sources of employment information. Students who participate typically gain useful contacts and take part in important networking activities that are connected to potential employment prospects (Monahan et al., 2014).

2.6.2 Networking. Much research has focused on networking as a job search method (Van Hoye at al., 2014). Sanyal and Neves (1998) described networking as a process that takes place in three stages for both personal and professional purposes: a) planning, b) connecting (i.e., making contact), and c) sharing information. In order to make connections through networking, one must have access to a certain amount of what social scientists have termed "social capital," which is composed of the informal or formal contacts in one's network. Ben-Porath (1980) described social capital in terms of one's "F-Connections"—families, friends, and firms—which

act as sources of information that could be useful in the job hunting process (Try, 2005). Reaching out to social contacts, as a job search strategy, frequently enables graduates to gain employment that is related to their educational background, such as their university degree qualifications (Franzen & Hangartner, 2006). Moreover, this informal referral system is quite commonly practiced all over the world by both jobseekers and recruiters (Franzen & Hangartner, 2006; Scheele & Madison, 2010). Pairing prospective employees with available positions in this way is especially useful for employers, as it results in no direct costs for them.

The relative inequities in the distribution of social capital are reflected in several ways: access to positions along gender lines and the distribution of qualifications across degree-level educational attainment and within many professions. A good example of this is the maledominated field of engineering in which females have less access to job-procuring social capital than men (Try, 2005). Thus, the distribution of social capital influences informal search methods through weak ties (Try, 2005).

While many people discover job vacancies and are hired for positions through informal job searches such as contacting friends and relatives (Cahuc & Fontaine, 2009), the effectiveness of informal networking does not necessarily depend on how many times a person reaches out to their contacts (Wanberg, Kanfer, & Banas, 2000). For students, networking through informal sources is frequently unsuccessful because of their comparatively limited professional networks. However, jobseekers can expand their professional networks by using social media (Sułkowski & Chmielecki, 2014). With this in mind, research affirms that "networking does indeed lead individuals into new jobs" (Wanberg et al., 2000, p. 501).

Research by Sin and Kim (2013) found that international students use social networking sites as a means to access information on job vacancies rather than building social capital and

networking. Building social capital is especially important for international students. On-the-job training opportunities (i.e., internships and co-op programs) are particularly valuable in helping students further develop their professional networks (McKeown & Lindorff, 2011), in addition to giving them a better understanding of the context and requirements of working in their disciplinary area.

Ultimately, it is highly recommended that jobseekers exercise both in-person and social media networking. Both of these avenues are important for jobseekers (Kibbe, 2009), who are advised to focus some of their efforts towards expanding their personal and professional social networks (Silliker, 1993). Marmaros and Sacerdote (2002) found that students often consider networking through weak ties an essential part of their job search process. Despite its usefulness, it is recommended that one should not heavily rely on networking as a sole and exclusive job search method; rather, it ought to be accompanied with other methods (Wanberg et al., 2000).

2.6.3 Weak ties and strong ties. The research literature on job searching differentiates between "weak ties" and "strong ties." According to Yamkovenko and Hatala (2014), "In the job search context, weak ties could be acquaintances, contacts of close friends, recruiters, interviewers, and others with whom the job seeker does not interact regularly" (p. 114). On the other hand, strong ties include relationships with closer contacts such as family, friends, and relatives (Granovetter, 1973; Try, 2005).

Granovetter (1983), among others, pioneered the notion of strong and weak ties within one's social capital. He argued that these ties help people to access job openings in different ways. While weak ties can provide a person with wider access to resources and information, strong ties can be more effective and are often more readily accessible to the jobseeker. In one of his later studies, Granovetter (1995) noted that individuals were more successful in job hunting

when using their weak ties rather than their stronger ones. On the contrary, Yamkovenko and Hatala (2014) contended that weak ties may not, in fact, lead a person to job openings. They suggested that the chances of weak ties providing a person with detailed information about a prospective job are relatively low. It is noteworthy that nowadays social networking websites such as LinkedIn and others provide those looking for work with plenty of opportunities to build a broad network of weak ties alongside stronger ones (Archambault & Grudin, 2012).

While strong ties involve relationships with friends, family, and relatives, one's network of strong ties often tends to lack the broad reach of weak ties, which involves a vast degree of connections. Also, variations exist within each strong tie and weak tie network. For example, among one's strong ties, friends may be more influential than relatives because of their heightened diversity and broader access to job-related resources (Lin & Dumin, 1986).

2.7 Other Job Search Strategies

2.7.1 Internet job search. Direct application to potential employers is associated with a higher probability of attaining employment (Kuhn & Skuterud, 2002). Prior to the advent of widespread internet access and usage, people would send applications directly to employers by post in response to their advertisements in newspapers (Bradshaw, 1973; Holzer, 1986); however, nowadays one accomplishes the same task using the internet by applying for positions advertised on company websites (Bissola & Imperatori, 2013).

Use of the internet as a job search tool better enables those seeking employment to heed Royal's (2001) advice; that is, jobseekers should approach potential employers after they learn more about posted employment opportunities by acquiring more knowledge about a company and its jobs. The internet has changed the nature of the job market by making information about positions more accessible to jobseekers (Wanberg, et al., 2000). It has also bolstered the position

of employers by enabling them to utilize employee recruitment websites (Bissola & Imperatori, 2013), social media (Karlin, 2005), and their own websites (McKeown & Lindorff, 2011; Wilden, Gudergan, & 2010). Research on the use of online job search tools has shown that students, post-secondary graduates, and jobseekers in general use the internet to look for employment opportunities because it is both a convenient and affordable avenue for job hunting (Lin, 2010; McKeown & Lindorff, 2011; Wanberg, et al., 2000).

The internet has also made it easier for recruiters to reach a wider range of candidates who can provide companies with applicants who are best suited for job vacancies (Feldman & Klaas, 2002). Likewise, job applicants have the opportunity to search among a larger number of employers in a short time (McKeown & Lindorff, 2011), which can potentially increase their chances of finding job leads (Garge & Telang, 2011). Scheele and Madison (2010) recommend that jobseekers regularly update their online profiles and resumes in order to increase their chances of being recruited and thus successful in their job search.

2.7.1.1 LinkedIn. Using social media and online browsing for job hunting and recruitment are still considered to be secondary to traditional job search strategies; however, online search tools have become increasingly important in light of the relative ubiquity of personal computers and other devices equipped with internet connectivity (Bissola & Imperatori, 2013). LinkedIn is a professional social networking website (Bissola & Imperatori, 2013; Sułkowski & Chmielecki, 2014) used by jobseekers for career management (Hempel, 2014), job searches, tapping into weak and strong network ties for employment referrals, and contacting employers. In 2008, the total number of LinkedIn users was 75 million. In 2012, this number reached 100 million, (Archambault & Grudin, 2012) making it the most used professional social

media network for job searching and employee recruitment (Bissola & Imperatori, 2013). Most of LinkedIn's users are professionals who are already employed (Garge & Telang, 2011).

LinkedIn is an information-rich source for both jobseekers and recruiters. While it connects people with others they know, it also enables users to connect with people beyond their own contacts. By allowing users to network with the contacts of other users, LinkedIn widely broadens the potential employer pool and sources of information related to job openings (Hempel, 2010; Karlin, 2005). LinkedIn is also a useful and efficient tool for recruiters because it can serve to boost "the quantity and quality of candidates" (Sułkowski & Chmielecki, 2014, p. 72). A survey by Adams (2013) showed that 97.3% of a sample of 1,848 staffing professionals used LinkedIn as a recruiting tool in 2012.

2.7.2 University career centres. Career and employment centres within post-secondary institutions are one of the most essential resources for assisting the population of post-secondary student and graduate jobseekers in their job search process (Bissola & Imperatori, 2013; Gerard, 2011; Marmaros & Sacerdote, 2002; Ng & Burke, 2006). These career centres offer a broad range of career preparation services (McKeown & Lindorff, 2011; McIlveen & Pensiero, 2008) with the most popular services among students being resume writing assistance and interview preparation workshops (Gerard, 2011).

In one major study of career services in higher education, Dey and Cruzvergara (2014) examined university career centres and how they have changed over the past century. The researchers argued that university career centres have tended to focus on traditional ways of delivering content and information to jobseekers rather than helping to create and facilitate connections with the outside community. They further highlighted the importance of partnerships between their services and personnel, faculty, and departments within the university. They also

emphasized the need for partnerships between university career centres and industry. The importance of partnership between on-campus career centres and academic departments has been echoed in case studies by Barthorpe and Hall (2000) and McIlveen and Pensiero (2008). Strong partnership with industries is highlighted by McIlveen and Pensiero (2008) in particular as central to successful graduate transitions to the labour market.

Dey and Cruzvergara (2014) recognized that career professionals on campuses play a central role in providing both students and employers with information and resources they need for successful job searches and recruitment exercises. They perform a key responsibility in helping to focus student and graduate jobseekers who often struggle with internet information overload. The researchers singled out Stanford University, The College of William and Mary, and Miami University as successful campus career centre innovators. They pointed out the more modern career service practices that have been put in place at these universities, such as having more informal group sessions with students across campus rather than "the traditional workshops and presentations" (p. 10). Successful career centres such as these institutions also tend to utilize technology to engage and motivate students rather than simply providing information in the traditional sense. Dey and Cruzvergara (2014) advocate better use of technology and career service approaches that are better tailored and customized to meet the individual needs of jobseekers.

2.8 Factors Influencing Choice of Job Search Strategies

Much has been written about the factors that influence one's choice of job search strategies (Bradshaw, 1973; Huffman & Torres, 2001; Marmaros & Sacerdote, 2002; Nikolaou, 2014; Shantz; Wright, & Latham, 2011). In the following section of this chapter, these are clustered into primary and secondary factors influencing specific job search methods. The

primary factors are those that are prominent in the research literature. They are further categorized into the following five main factors: a) race, b) age, c) gender, d) education, and e) social status. The remaining secondary factors are comprised of influencing factors that have tended to receive less attention or emphasis in the research literature, such as information richness, cost, convenience and usefulness, and personality traits.

2.8.1 Primary factors.

2.8.1.1 Race. Bradshaw (1973) explored different job search methods used by jobseekers of different racial backgrounds, namely Caucasian and African-American jobseekers. He found that Caucasians used newspaper wanted ads and friends and relatives more than African-American workers in Philadelphia's inner city. In comparison, African-American workers tended to use state employment agencies more. In another study, Johnson (1997) looked at race and gender together and found that African-American women and men and Caucasian women utilized formal job search methods more than Caucasian men. The latter tended to use informal search methods more than all others. She also found that Caucasian women were more likely than Caucasian men to use a career centre.

2.8.1.2 Age. In addition to racial factors, Bradshaw (1973) also examined the methods used by jobseekers of different ages. This research showed that jobseekers between the ages of 20 and 24 used public and private employment agencies and newspaper ads more than teenagers. Teenagers tended to favour approaching employers directly. A more recent study, which also took the use of internet job search techniques into consideration (Nikolaou, 2014), found that younger jobseekers tend to use online social networking websites more in their search for employment.

2.8.1.3 Gender. Studies that have accounted for gender differences in job searching have consistently found that men are prone to approach their immediate contacts more than women (Bradshaw, 1973; Huffman & Torres, 2001; Marmaros & Sacerdote, 2002; Shantz, Wright, & Latham, 2011). Men are also more likely to contact potential employers directly (Bradshaw, 1973). This is significant in terms of information flow and access to resources. In lesser contrast, Wei-Cheng and Kopischke (2001) found that male and female college students used networking for job hunting more equally.

Kimmell et al. (2007), Huffman and Torres (2001), and Wei-Cheng and Kopischke (2001) reported that females prefer to use newspaper advertisements for job searching more than men. In studies by Bradshaw (1973), Huffman and Torres (2001), and Shantz et al. (2011), it was concluded that women jobseekers are less likely than men to rely on employment agencies, campus career centres, or labour unions as potential leads for job opportunities. On the whole, studies such as these provide some evidence to suggest that women tend to select some of the more passive job search methods compared to men (Grenon, 1999). On the other hand, research also shows that women tend to use social networking websites and online job boards more than men. Interestingly, men have been shown to make more use of LinkedIn than do their female counterparts (Nikolaou, 2014).

2.8.1.4 Education. A variety of studies have examined the influence of education levels on job search strategies. These have looked at the selection and use of different job search methods and their relationship to the educational background of jobseekers themselves as well as their parents. Johnson (1997) compared one's preference for or choice of formal and informal job search strategies and found that those who are less educated tend to use informal methods

more often, whereas jobseekers with higher levels of educational attainment tend to use formal methods more frequently.

Those with lower educational attainments rely more on networking as a job search strategy (Try, 2005), while more highly educated individuals tend to use networking less in job hunting (Monahan et al., 2014). Chua (2007) observed that less educated jobseekers were three times more likely to contact friends and family in their search for employment. There is some disagreement on this, however; Van Hoye (2013), for example, has argued that it is actually more likely for those with higher level educational qualifications to use referrals and informal job search strategies.

Kuhn and Skuterud (2002) found that individuals with higher levels of educational attainment tend to use the internet more to look for a job. In fact, online networking through LinkedIn has been shown to be more favored among jobseekers with higher level educational qualifications (Nikolaou, 2014).

The educational attainment of parents has also been examined in the context of their influence on jobseekers' employment prospects and job search approaches. For example, it has been observed that if a jobseeker's parents have the same level of educational attainment as the jobseeker themselves, they are more likely to open the doors of opportunity for the jobseeker though their own networks, knowledge, and prior job search experiences (Try, 2005).

Try (2005) also investigated job search behaviour from the perspective of degree-specific educational preparation. This study found that a lack of employment opportunities for law graduates influenced them in their choice to use public employment services more frequently. In contrast, other professional degree graduates who graduated from business administration programs were more unlikely to use these employment services. It has also been suggested that

lower levels of educational attainment might be associated with increased use of governmentinitiated career programs (OECD, 1999).

Finally, one study of university graduates found that these job-seeking alumni used their campus career centre, online job search engines, and employment agencies more frequently than other approaches because they believed the reputation of their university and their own educational background would be more influential and better recognized if they chose these routes in seeking out potential employers (Bissola & Imperatori, 2013). Also, students who use the university's career services are reported to land more prestigious jobs (Marmaros & Sacerdote, 2002).

2.8.1.5 Social Status. Nasser and Abouchedid (2006) asserted that post-secondary graduates with lower socio-economic status make greater use of strong ties when they are seeking employment, including their network of family and friends, in comparison to graduates from higher socio-economic groups. They suggested that those from more affluent backgrounds are more likely to formally apply to a job. Try (2005) suggested that rather than take into account social status on its own, researchers ought to consider a combination of influencing factors such as educational background, academic performance, and socio-economic status. He suggested that given the choice of using ads and career centres as job search methods, those from lower socioeconomic strata may be more likely to use ads, as they tend to be freely available and easier to access. Interestingly, Nasser and Abouchedid (2006) found that public university graduates used informal methods like contacting family and friends more frequently than graduates of private universities, who used formal job search methods more often.

2.8.2 Secondary factors.

2.8.2.1 Information richness. According to Granovetter's model of job search, certain occupations expose individuals to higher levels of information through their work contacts. Kim (1994) argues that this "information richness is an important component in explaining job search method choice" (p. 98). Thus, more advantageous jobs provide greater human interaction and increased numbers of personal contacts, which in turn may potentially lead to more work opportunities through these networks of contacts.

2.8.2.2 Cost, convenience, and usefulness. It is argued that informal sources of information such as personal contacts and other informal networks result in lower costs to jobseekers (Franzen & Hangartner, 2006; McKenna, 1985). However, the motivation to use these social networks often depends on whether jobseekers perceive them to be as useful and enjoyable as other alternative job search approaches (Suki, Ramayah, & Ming, 2011).

Perceived usefulness and ease of use are also factors that may influence one's choice of job search strategies. For example, jobseekers tend to use online job search strategies, which are both easy to use and frequently yield successful results (Lin, 2010). This has implications for online recruitment or job search websites, which provide the most user-friendly and convenient navigation features. Although the online job board Monster.com is a pioneer in the area of employment websites, some jobseekers believe that online job search engines are not a priority, because while sites such as Monster.com provide a wide range of job postings, they often do not enable users to connect with the specific jobs that they are looking for (Gunz & Peiperl, 2007). Among jobseekers, post-secondary graduates have reported finding it easier and more efficient to browse through online job search engines in order to look for a job (McKeown & Lindorff, 2011). In searching for jobs on the internet, these jobseekers report they can access greater

amounts of information about job vacancies and can gain a better sense of the expectations of available positions.

2.8.2.3 Personality traits. It has been suggested that certain personality traits are connected to job search preferences and success in seeking employment. For example, extroverted individuals tend to be offered information about job openings more often through word of mouth because they do more networking (Van Hoye, 2013). This thinking is far from conclusive, however. Ellis and Taylor (1983) contended that employability resulting from strong network ties is more related to the social capital of a jobseeker rather than their social skills. A study by Royal (2001) found no connection between personality and social skills and the choice of job search methods. More importantly, he argued that when jobseekers approach their social contacts in search of a job, it is the social contacts, rather than the personality of the jobseeker, who provide increased opportunities. Other research has examined the effects of self-esteem on the choice of job search strategies. Ellis and Taylor (1983) reasoned that individuals with low self-esteem tend to utilize job search methods that have not necessarily been proven to be successful strategies. Research has also shown that students with low self-confidence and underdeveloped employability skills are more likely to seek out the services of publicly subsidized employment services (OECD, 1999).

2.9 Conclusion

This chapter has reviewed and outlined some of the major studies on the oil and gas labour market in Newfoundland and Labrador. It has also covered a variety of general job search strategies individuals use along with the factors influencing one's choice of job search methods.

Despite the volatility in international oil and gas markets, Newfoundland and Labrador oil and gas labour market projections suggest that there is a positive future ahead for those

seeking employment in the Province's oil and gas industries. With ongoing major oil extraction and production projects such as Hebron and Hibernia, it is expected that the Province will have a high demand for skilled workers in the coming years.

While this literature review provides important background information on job search strategies, the aim of this research study is to provide new knowledge on graduate students' choices of job search strategies as they attempt to enter Newfoundland and Labrador's oil and gas labour market. The next chapter on research methodology describes the author's chosen approach for examining how graduate students majoring in oil and gas-related degrees connect to employers and how oil and gas jobs are filled in the Province.

Chapter 3

Methodology

3.1 Introduction

This chapter presents the methodology of this study, which includes research design, data collection and sampling, data analysis, and ethical considerations. The individual research questions of this study are as follows:

- What job search strategies do international graduate students in the Oil and Gas Engineering and Earth Sciences programs in Newfoundland and Labrador use to look for jobs in the province's oil and gas industry?
- 2. What are the factors influencing oil and gas student job seekers' choices of job search strategies?
- 3. What supports, guidance, and resources do higher education institutions in Newfoundland and Labrador provide to students who are seeking employment opportunities in the oil and gas industry?

The provincial skilled worker shortage raises concerns for the already existing skilled and educated potential workers (i.e., international graduate students and the problem of their search for employment). As a result, this study aims to a) provide an understanding of job search strategies that international graduate students in the Oil and Gas Engineering and Earth Sciences programs when they search for employment within Newfoundland and Labrador; b) identify factors that influence students' choices of such job search strategies; and c) investigate the role of provincial post-secondary education institutions in providing services and support during the students' career search process.

3.2 Research Design

The central phenomenon of this study is "student job search in the oil and gas industry." This phenomenon encompasses the choices that students make in using specific search strategies, factors that influence their choices, and their use of the university career centre in their search processes. The inquiry into the central phenomenon requires a research approach that gives an in-depth insight into the lived experiences of the participants. A phenomenological qualitative approach maintains such qualities and was selected for this research project because "the intent of qualitative research is to understand and explore the central phenomenon" (Creswell, 2012, p. 130) through detailed analysis of the data.

The epistemology in this qualitative approach enabled the researcher to investigate the knowledge construction of the participants who have experienced or are experiencing the phenomenon by probing into their behaviours, beliefs, and feelings. In such a qualitative approach, "knowledge ... rests within subjective experience" (Morgan & Smircich, 1980, p. 497) of the participants and is presented in descriptive responses to interview questions. The subjective epistemology of the qualitative approach contrasts with positivist epistemology, which is more concerned with the relationship of cause and effect between the elements of the study (Morgan & Smircich, 1980). A phenomenological qualitative approach was suitable and selected for this study rather than a quantitative one, and it is intended to provide greater insight into student job search strategies and expand on related research.

3.3 Data Collection and Sampling

In a qualitative study, interviews are a key means of data collection that enable the researcher to gain insight into the experiences of the participants and see the phenomenon of the study from the participants' perspectives (Lester, 1999). Having said that, an understanding of

how international graduate students in oil and gas-related programs in Newfoundland and Labrador are seeking employment required an emic (or insider) view on their experiences. The research is guided by a holistic approach to data collection using in-depth, semi-structured interviews and "the external world is reduced to the contents of personal consciousness" (Groenewald, 2004, p. 4). Exploring the "external world" guided by Husserl's phenomenological approach, semi-structured interviews provide greater flexibility for the researcher to delve deeper with follow-up and open-ended questions and allow researchers to enter "the psychological and social world of the respondent" (Smith & Osborn, 2003, p. 59) by giving the researcher the opportunity to exceed the limitations of a structured interview method. On the other hand, this method of data collection also gives more voice and authority to the research participants and reduces chances for the researcher to impose subjectivity. As a result, there is a better chance for emerging themes to rise to the surface in semi-structured interviews.

Moreover, interviews are a means of self-discovery through which the participants can convey their meanings to the researcher on the phenomenon under inquiry (Dilley, 2004). Indepth interviewing is a method in qualitative research by which previously unknown knowledge is conveyed to the researcher. Hence, in-depth interviews, as a method of data collection in qualitative research, create and negotiate knowledge (Legard, Keegan, & Ward, 2003). Legard et al. (2003) state that knowledge construction through interviews in qualitative research takes place "through collaboration between the interviewee and the researcher" (p. 140).

This is to say that one-to-one and face-to-face interviews (Creswell, 2012) with students as jobseekers provided detailed insight into their experiences regarding their uses of job search strategies. To explore factors influencing the choices of specific search methods, semi-structured interviews helped to facilitate further questions towards understanding what influenced students'

decisions to choose their own strategies in instances where they were unable to identify the influencing factors on their own. Also, this approach enabled the participants to speak more freely in expressing their views on the career services they have used and give recommendations for change. For this research project, a questionnaire with thirteen questions was prepared and the items were checked to reflect the research questions. Student and graduate responses were audio recorded by the researcher and these data were subsequently transcribed for analysis by the researcher.

Another method of data collection utilized for the purposes of this study was document analysis. This included an analysis of reports and documents from the public sector of the oil and gas industry, post-secondary education institutions, and Statistics Canada, which are relevant to this study. These reports included secondary documents such as "public document sources … census reports, statistical bulletins, departmental annual reports … and private documents from private sector businesses" (Gaborone, 2006, p. 223). However, it is noteworthy to mention that only publicly available documents were reviewed for this study.

In qualitative research, participants are selected based on the purpose of the study (Whitt, 1991) and the context of the studied phenomenon. The local oil and gas labour market, as the context, gives us the background information about the behaviours of the participants (Berg & Lune, 2004). For example, in this study, knowledge of the provincial oil and gas labour market, presented in Chapter 2, helps us to better understand the feelings and attitudes of the participants in their responses. Memorial University of Newfoundland was selected as the site of the study because it is the only university in the Province of Newfoundland and Labrador, the location where the researcher is based, that provides Oil and Gas Engineering and Earth Sciences programs at the graduate level. The goals of this study required selecting a sample that best

represented the characteristics reflected in the intent of the research questions. With the above in mind, a specific group of graduate students at Memorial University of Newfoundland were sent a study recruitment invitation (See Appendix C) via an email from the university's School of Graduate Studies. This call for research participants indicated that the inquiries in this research project were delimited to international graduate students who either were in their final year of a master's degree or were studying in their third or final year of doctoral studies. Also, the programs of interest to this study were Oil and Gas Engineering or Earth Sciences. To reflect the distinct qualities of the population under study, the sampling method used for data collection was homogeneous sampling. To have a homogeneous sample, participants cannot be randomly selected, and thus sampling is purposive (Patton, 1990). Purposive sampling as a non-probability sampling method is aimed at a specific population; as a result, it provides better insight into the investigated phenomenon (Creswell, 2012) and helps locate more potentially information-rich participants (Patton, 1990). Moreover, snowball sampling, whereby the researcher asked respondents to give referrals to other possible respondents, was also used later in the interview process to help locate, identify, and recruit additional participants (Creswell, 2012). Snowball sampling is mainly important because it enables the researchers to expand the "sample beyond contacts known to the researcher" (Lynch, n.d). By using this method, the researcher was able to leverage the network of three participants who responded to the "call for participants" email to identify five additional participants for interviews. Furthermore, to gain better insight into the services offered to students and to learn about the programs and services offered specifically to the targeted population of this study, the recruitment message was sent to staff members at Memorial University's career centre (See Appendix D for recruitment script). This resulted in the recruitment of two additional participants who agreed to take part in interviews. Conversations

with the career centre also provided a better understanding of and more information about students' access to the services available at the career centre. Therefore, in this case, expert sampling helped identify individuals (Patton, 1990) with expert knowledge of career services and programs at the university.

The research participants for this project were mainly students, and interviews with career centre staff were only conducted for information purposes (See Appendix F for administration interview questionnaire). This is because the focus of the study was to investigate research questions from students' perspectives. There were eight student participants in total, referred to as "participants" hereafter: five master's students and three doctoral students. The sample size was ideal for this study because the research experienced a level of data saturation by the sixth interview. The master's students included two females and three males, all in their final year of studies. One of the female students was pursuing a Master's in Earth Sciences and majoring in geology (Student A, MSc) and the other was in Oil and Gas Engineering (Student B, MEng). Among the three male master's students, one was enrolled in Oil and Gas Engineering (Student C, MEng) and the two others (Student D, MSc and Student E, MSc) were studying earth sciences (geology). The doctoral students included one female in Oil and Gas Engineering (Student F, PhDEng, in her final year of studies) and two male students: one in his final year of Oil and Gas Engineering (Student G, PhDEng) and the other (Student I, PhDSc) in his third year of Earth Sciences (Geology).

3.4 Data Analysis

Participants in this study were asked to respond to 13 questions that reflected the research questions. The responses provided by study participants were recorded and these interview recordings were transcribed.

A detailed and thorough analysis of the interview responses was carried out in an effort to reflect participants' feelings and experiences. This involved interpreting and making sense of the answers of the respondents (Smith & Osborn, 2003). Interpretive analysis was used to place the data in the appropriate context of the study, in other words, to investigate the central phenomena contextually (Larkin, Watts, & Clifton, 2006). For this interpretative analysis, the whole transcript for each research participant's interview was treated as data that could be used to learn about the perceptions and feelings of the participant. The researcher identified commonalities and related words and phrases and these were clustered and coded. As is frequently the case in this sort of phenomenological analysis, there was a possibility that clusters of data would overlap with one another (Hycner, 1985).

At the beginning of the data analysis, the interview transcriptions were treated as a whole dataset as the researcher attempted to understand the experiences and feelings of the respondents in the research. While the data set was studied, the researcher took notes to inform the later stages of the data analysis. Then, in the later stages, the data was coded and categorized into groups of emerging themes. The coded data was constantly compared to data groups until all the data was categorized. In this process, data that did not contribute to understanding the phenomenon was omitted from the analysis. Later, coded data in each group was grouped into subcategories. This is an inductive method of data analysis in qualitative research; it helps the researcher to obtain more specific results from the initial general analysis. As a result, a second reading of the data was carried out with the aim of checking the initial annotations against recurring concepts. This revealed three general groups of data based on research questions and two emerging sets of data.

The three general groups of data included a) job search strategies that students use or have used, b) factors that influenced their choices of using such job search strategies, and c) university career centre programs, supports, and services that students availed of in their search process. The two emerging sets of data included a) job searching as a planned project and b) students' perceptions of the oil and gas labour market of the Province.

3.5 Ethical Considerations

For the purpose of the interviews in this study, ethics approval was sought and received from the Interdisciplinary Committee on Ethics in Human Research of Memorial University of Newfoundland (ICEHR). The materials reviewed by the ICEHR included a research proposal, including the draft interview questions and recruitment letter scripts employed in carrying out this study. As per the ethical requirements of the university, students were informed about the benefits and risks of their participation in the study and the objectives of the study. This information was outlined in the consent forms approved by the ICEHR prior to commencing the research. All student participation in this study was voluntary and they were asked to inform the researcher in case of their decision to withdraw. To date, no participant has requested to withdraw from the study.

Also, interviews with the university career centre's administration helped the researcher gain deeper insight into the services provided to students. These data were not intended for use in in-depth analysis; rather, it served as background knowledge that enabled the researcher to better understand students' responses that were relevant to the third research question regarding career centre's services and programs for job-seeking students. Nonetheless, consent forms were provided for members of the career centre administration as well (see appendices A and B for a

copy of students' and administration's consent forms, respectively). Copies of all consent forms and recruitment scripts are provided in the Appendix section of this thesis.

3.6 Summary

This is a qualitative study with a phenomenological approach. The chosen methodology enabled the researcher to best investigate the central phenomenon by using qualitative methods that gave insight about the lived experiences and feeling of the study participants. As a result, a purposive homogeneous sample was used to gather data on the investigated phenomenon. In addition, after identifying the themes reflected in the research questions, interpretive analysis was used to analyze the data for additional emerging themes. The data are presented in Chapter 4 of this thesis along with an analysis of selected quotes from the interview respondents.

Chapter 4

Findings and Analysis

4.1 Introduction

This chapter will report the data generated by the semi-structured interviews and postinterview informal conversations with participants. The data generated from interview transcriptions were coded, categorized, and themed based on the research questions. At the end, categorization of the data resulted in two sets of key findings and interpretative data analysis, subsequently resulting in two emergent data sets. The following section presents the findings of this study based on the data analysis carried out. One of the research questions explored for the purposes of this study addressed the job search strategies that students used the most. The two most commonly used approaches were networking and online job searches. Students preferred to network in searching for employment because it enabled them to a) be face-to-face with employers, b) learn about employers' expectations, c) make and maintain contacts, and d) improve communication skills. At the end, recommendations from students for the university career centre are provided.

Students' online job search strategies commonly included four main sources a) LinkedIn, b) job search websites, c) company websites, and d) professional associations. A variety of factors influenced students' choice of job search strategies: a) the type of job sought, b) peers and word of mouth, c) the type of industry in which the individual was interested, d) whether or not the person had previous work experience, and e) if the individual had assistance from the campus career centre.

Two overall emergent themes were identified to exert an influence on one's choice of job search strategy. First, for a number of participants in this study, their strategy for seeking

employment was contingent on whether or not the job search was a planned project included in their program of study. International graduate students' perceptions of the characteristics of the oil and gas labour market of the Province was also an overarching influence on student and graduate employment choices.

The presentation of data that follows is accompanied by verbatim quotes from respondents, giving voices to individual participants. These direct quotations from participants are provided in order to enable the reader to better understand the studied phenomenon. These quotations illustrate individual explanations and present examples that provide a better understanding of student job search strategies and choices.

4.2 Job Search Strategies

Students were given a list of possible job search strategies and were asked to choose the strategies that they used and/or had used. They were also asked to indicate the ones they thought were the most effective (See Appendix E, question 8 for the complete list of job search strategies). The analysis of interviews with graduate students revealed two major job search strategies: networking and online job search. These two strategies were the ones respondents claimed to have used, preferred to use, and/or were considered to be the most effective for finding a job.

4.2.1 *Networking*. The data collected for this study show that students tend to define networking in their own terms and have a clear understanding of the objectives of networking activities. They also set their objectives, pick the players for their networking plans, and seek opportunities to access key social contacts. Students frequently defined networking as making contacts through their friends, teachers, and university supervisors; however, making connections was not limited to who they knew. The data from the interviews show that students

feel strongly about reaching out to friends and social contacts to make connections that would lead them to potential employment opportunities. One student referred to such leads as "direct contacts," who were perceived as individuals who could lead them directly to human resources (HR) professionals. For example, Student C, MEng said, "through friends and contacts, yeah I think these are the most effective in my experience to get direct contact ... contacting directly generally means HR people."

For many jobseekers, making connections means looking for avenues to connect to people and to make focused efforts to build relationships. Students have the knowledge of where to look to build a network of social contacts, whether making friends or acquaintances. Student A, MSc noted, "my personal contacts, like through the course of my studies, say my profs or other people I came in contact with. I'll also contact them personally."

The data collected shows that networking, in the way it is practiced by the students who were interviewed, is also a strategic plan that requires clear objectives. A strategy in approaching potential contacts was shared by Student F, PhDEng:

I would do that personally, just to ask, not anything like pushing them but just by friendly asking them, do you guys know if the company is recruiting? Do you know the best person to contact? Here is a copy of my CV, if you want something, please place my CV there. And this will be done to those people who I am comfortable with in communicating in such a way. I will also exercise those avenues, yeah [*sic*].

Student E, MSc states, "Sometimes you have to bump into someone and they mention that, and I take it as a connection, I think that actually matters even more than just randomly sending out your resume here and there." This response suggests that building connections can extend to any possible opportunity that connects jobseekers with potential leads on job opportunities. This may

happen by randomly meeting people and learning that they have industry-related information. But, as pointed out by Student E, it is the effort of the individual that can turn such encounters into opportunities. The statement also suggests that this student values direct encounters more than going through the formal motion of submitting applications to potential employers.

The interview data showed that the respondents often made focused efforts to expand their social capital by attending networking events. There appears to be two major players who moderate students' access to information about networking sessions. One of these is the university career centre where students reported attending organized events often, such as employer information sessions and career fairs. The other major player is professional organizations, which were recognized mainly by students who had previous work experience. In many cases, students attended company events that were advertised through such organizations. These advertisements were frequently received by students via email.

When students were asked about their networking activities, including the reason they networked, all of them believed that networking is the optimal way to look for a job because a) it enables them to talk face-to-face with employers and expose their skills; b) they can learn about employers' expectations by asking them directly, and as a result they have a better idea about what skills they need to improve; c) they are better able to make and maintain contacts, with the aim of eventually approaching potential employers when they start to actively apply for jobs; and d) networking with employers was regarded as a means of improving communication skills, which they believe employers value greatly.

1) Networking face-to-face with employers. At Memorial University, students have many chances to meet various employers through events held by the university career centre. One of these opportunities is the career fair, which enables students to oftentimes talk directly to

employers and their employees. At these job fairs, there is usually one human resources staff member available to answer jobseekers' questions. According to study participants, most companies encouraged students to visit the company's website to learn more about their projects and job openings and to submit resumes or create a personal profile on the company website. Through direct networking with employers at career fairs, students learned about companies' processes for submitting applications as well as other expectations, such as providing copies of transcripts or completing an initial screening quiz. Student D, MSc stated the following:

Of course I attended the job fair in September. It was a good opportunity for us to talk face-to-face about the job opportunities. So if they got a job, they recommended us to go online and upload our resume/cover letter.

The interview data indicates that when students go to networking events, especially a career fair, they have particular goals in mind: "When I went to a career fair, I was looking both ways, information for my research because it is related to the oil industry and also any potential job opportunities that you can get" (Student E, MSc). Students who attended job fairs often engaged with targeted employers at booths representing a wide range of companies. However, despite their overall satisfaction with their career fair experiences, some students mentioned that their objectives did not match with employer's goals. In one case, Student F, PhDEng perceived the intentions of some employers to differ from students' objectives:

Some employers try to promote their company rather than actually hire or look for some candidates in the job fair. I feel that they are not really interested. When I approach them and say that ok, is there any posted position for R&D⁵ position, they say ok, check our website, so that's kind of general and typical answer.

⁵ Research and Development

Another networking experience described by students was "speed networking," where they are provided an opportunity to sit briefly with an employer and take a few minutes to talk and exchange information. This way, students have a chance to meet with a variety of different employers in a short period of time. There was one student who did not have previous industry experience and had not attended the event; therefore, his comments were based on his friends' feedback. Student B, MEng was interested to try methods that would give her better search options: ones that are not only complementary to other search methods but would also act as mediums for job leads.

I heard there are some social activities like bringing people from the industry for the students; I heard that one last year, so you have three minutes to talk to one person ... I am interested to attend next time because the problem I have now is like [*sic*] all I have in my hand, all the resources I have, is the online application system! That's it! I can't see the person there, I can't see the face.

These comments also show that although the student had not attended the networking session, her friends' feedback helped her to reflect on and see the value in attending such events. This student, like a number of other respondents, emphasized the importance of face-to-face conversations and exposure of personal skills to potential employers. Moreover, she is seeking to diversify the tools she is relying on for her job search and is willing to try the approaches her peers viewed as effective.

Students have broad views on networking. In fact, any opportunity that would provide exposure to the labour market and the world outside the university would be considered networking. However, Student G, PhDEng has made multiple efforts to network with a specific

objective in mind. The student perceives networking as an opportunity to learn about the labour market and gather preliminary information at the beginning of a multi-stage job search process.

I try to attend different job fairs both here at MUN⁶ and other professional agencies, such as NOIA⁷ or a couple of engineering associations. I usually attend job fairs held by these organizations. Also, I try to attend as much networking sessions as I can. My purpose is to benchmark the job market here.

Organized networking events give students a chance to meet industry insiders; this is particularly important for students in research-intensive programs who see networking as a chance to share information about their own research projects that are in some instances sponsored by oil and gas companies. Student D, MSc indicated that Suncor Energy sponsored his project, and he had presented his work to company officials. He was pleased with this opportunity to network and believed that because he was the only international student who had presented to the company, there was a good chance the company officials would remember his work.

My project is based on Suncor, ... I did a presentation for them, if it's networking, yeah, if it counts, because I know some guys there, and some guys know my name because I'm the first, I think I'm the first foreigner to do presentation for them in their conference room.

Student C, MEng also had a positive networking experience:

I think since the start of my master's I started networking because my research is related to the oil industry here in St. John's. So since I started my research project here, I got more connections here, from the first month that I came here to Memorial.

⁶ Memorial University of Newfoundland

⁷ NOIA-Newfoundland & Labrador Oil & Gas Industries Association

As these data indicate, students highly value opportunities to interact personally with employers or other industry officials. This often provides opportunities for them to talk about their research projects or to make inquiries directly with potential employers. Networking is considered one of the most effective tools for their job search and events that provide opportunities to network are thus highly valued.

2) Learning about employers' expectations. In face-to-face networking, students have a chance to learn about what employers have to offer. They can also inquire about the qualifications needed for various positions and projects and learn more about the skills they need to improve on in order to match their potential employer's criteria. Student B, MEng provides some insight into how important it is for students to find ways to stand out in the labour market and to be competitive with their peers:

Talking face-to-face to potential employers is the best way to see what's the expectation from a fresh graduate and by talking to those people I would be able to identify my shortcomings and identify those aspects or those skills that I need to work more on before graduation, you know, to prepare myself and to make myself competitive in the job market, so that's the reason why I really use that strategy, you know, attending job fairs, talking to employers and the HR.

Similarly, Student G, PhDEng made the following observation to the researcher: I think attending the job fair is important because you learn more about them [the employers and companies] ... I think that's the best way to do, it is more directly to target or to see what the company wants.

In many cases, international students seeking to stay in Canada and work have the additional task of trying to better understand the Canadian labour market and the expectations of new graduates.

In some instances, they consider networking with employers to be the best way to learn about the Canadian labour market needs and the work culture. Student I, PhDSc demonstrates this way of thinking:

I attended the job fair to ask or understand how it happened here because the job market I looked for was in my country, which is quite different than here in Canada, at least I assumed it was different and my objective was to go and ask them to see what they look for, what priorities they have in terms of candidates and those things.

As a student with management experience, Student G, PhDEng further illustrates that networking is considered a vehicle for gaining labour market information that the student jobseeker can then use to prepare for the active job search after graduation.

My purpose is just to get familiar with the requirements of the potential employers especially here in Canada, what's their expectations from fresh graduates compared to back home because ... the criteria is almost the same but sometimes there are different focuses on the specific skills here in Canada compared to back home so right now I'm considering reading the employers' websites, attending job fairs or attending networking sessions.

3) **Making and maintaining contacts.** The students interviewed for this study demonstrated an awareness that networking is a lengthy process that is aimed, in part, at building one's contacts. They also believe that networking is a tool that aids in their job search process, but because networking can be challenging, many of them felt it was essential to start networking early in their studies. For example, Student A, MSc mentioned that he had already made a list of contacts.

I think when you are ready to go for job search, you probably have the tools already in your hand, what I mean to say is that I need contact. When I decide now to look for a job, I can't start building contacts now because it takes some time, so networking has to be a long-term process, for several months or years or a few events, so you'll get a certain number of contacts. When you start job searching, you use the contacts you build over years to use for your purpose so the tools I think, and especially for me, I think the tools already exist there and built up and when I decide to go for a job search, I just use the tools.

This quote demonstrates that Student A, MSc is prepared and confident in his job search process because he is aware of the time-consuming nature of building contacts through networking.

Two of the study participants (Student G, PhDEng and Student E, MSc) pointed out that a significant factor in networking is the actual selection of contacts. Consequently, in order to build meaningful and potential contacts, these students aimed to identify contacts who would be the most likely to direct them to job opportunities. Building a useful network of contacts requires an appreciable allocation of time, so some of the students made efforts to expand their social network early in their degree programs. Students also said they planned to touch base with their list of contacts in their final semester, when they are closer to graduation, to inquire about potential job openings and if their contacts are aware of any job opportunities.

Furthermore, Student E, MSc suggests that social media is another avenue for making and maintaining a list of networking contacts. Social media can enable jobseekers to identify potential contacts, and since contact information is readily available online, it may be easier than other search mechanisms to quickly find people with affiliations to specific companies.

When I network, especially through social media, I try to keep the potential contacts that I think might help me in future and are also related to my major, and then I disregard the ones that do not match my job search criteria. (Student E, MSc)

4) **Networking to improve communication skills.** One means of networking mentioned by Student D, MSc was volunteer work. Specifically, he believed that students need to develop their soft skills in addition to technical capabilities and volunteering could help improve communication and teamwork skills. It also helps to better adapt to the local culture.

Besides the technical skills that we learn from university, they [employers] need other skills like team work, communication, specific culture; different companies have different culture. That's more effective ... it builds confidence to communicate with people and know the local people and local community ... The communication skills totally help, and team work skills, yeah [*sic*], those are two important things I learned from volunteering. (Student D, MSc)

Interpretations of data on networking provide results on students' job search objectives and expectations, how they define networking, and what they consider networking; all of these are presented in the discussion section of this thesis.

4.2.2 Online job search.

4.2.2.1 LinkedIn. When I asked the students about the use of social media in job hunting, all of them mentioned that LinkedIn is a good way to keep in touch with the oil and gas companies and the HR professionals. There is ample support from the data that LinkedIn is considered an influential networking and job search tool. Some students with prior work experience used LinkedIn more often and were more aware of its different features. Jobseekers can join professional groups and company profiles to learn about their vacancies and projects and

to connect with industry people. Networking through social media is viewed by students as more streamlined in terms of identifying potential contacts: "Normally I go on LinkedIn daily, to see which connections I can get, and sometimes go directly to the company profile to see what they have to offer" (Student B, MEng).

The availability of various professional groups on LinkedIn provides ease of access to industry information. It is free to join company profiles and make connections with employees and employer, engage in group discussions, and search and access people, jobs, and companies. "LinkedIn is one of my top choices when I want to look for a job because you can get many features out of it. You can search jobs, join company profiles and other related groups" (Student A, MSc).

Students with prior industry experience were aware of the importance of keeping their LinkedIn profiles updated. As a result, two of the experienced students who carefully maintained their profiles (Student E, MSc and Student G, PhDEng) mentioned that they received job offers either directly from the employers or by head hunters through LinkedIn. "In my LinkedIn account I received just four job position offers and out of those, two of them was really serious [*sic*], they wanted me to start the job right away" (Student G, PhDEng). This is a success story of using LinkedIn, which could be valuable information for peers. A trusted network of friends carries information regarding trials and errors of their job search experiences. Student jobseekers who are aware of such success stories and are interested to learn about their friends' and their networks' experiences know that in order to be on top of labour market competition, they need to keep their LinkedIn profiles tailored and up-to-date:

Whether it's a research career or academic career, I'll probably try to build it [my LinkedIn profile] or modify it depending on where I am looking for and then I think I

will attract more potential employers because they also always keep looking for people in LinkedIn so if I orient my LinkedIn in a certain direction then I'll attract more potential employers. (Student I, PhDSc)

The data also shows that using LinkedIn goes beyond job searching. Students understand the value of being on track with the news and information flow in the industry, so they join LinkedIn groups in order to have a current mindset about the oil and gas market. LinkedIn is a networking tool used in the preliminary stage of the job search process.

I'm also using LinkedIn. On a weekly basis, I receive emails ... from LinkedIn groups. I have joined related associations, or the working groups related to the work I'm doing. I'm receiving notifications from those working groups to keep up-to-date with the changes in the industry. (Student C, MEng)

Data from students related to using LinkedIn also depended on where they are in their search process. Student I, a third-year PhD student in oil and gas engineering, revealed his preference for a passive job search strategy, where he only has a profile on LinkedIn for the time being:

I created LinkedIn when I came here ... it's a very general profile showing what my project is, the thesis title of my PhD and where I work, and a little bit of job responsibilities like bullet points, not too detailed, and my other experience, my volunteer work, and my interests. So it's a snapshot of my profile but it's not designed for a particular industry.

This approach was also taken because he had not yet determined his career options and still must decide if he can acquire an academic position or return to industry; however, the student is aware that once he decides his career path, he will need to customize his LinkedIn profile to match his desired career option.

4.2.2.2 Job search websites. Another pool of data focusing on using LinkedIn showed that students compared LinkedIn with online job search websites. Students preferred any search engine that would provide them with targeted information in a short amount of time. Job search websites that provide a wide range of jobs are regarded as ineffective because oftentimes search results do not match with what they are looking for, whereas LinkedIn provides a better-customized search engine. This view of LinkedIn as the superior tool was voiced by Student F, PhDEng, "In my opinion, LinkedIn is much more powerful compared to Indeed.com so I prefer to ignore job agencies, because I'm receiving much more related job opportunities." This increased rate of success might be because companies have an active presence on LinkedIn and post their vacancies on it.

The notion that general job search engines such as Indeed.com send more unrelated job alerts was echoed in other students' responses as well. This view, however, comes to light because of students' familiarity with LinkedIn and its features. Their successful experiences with LinkedIn could be a reason they compare the two.

Based on my experience with Indeed.com, I have been receiving a huge amount of emails. I can say at least two emails per day from those job agencies and I can say out of ten emails, maybe just one was related to my field, so in my opinion, I'm not a fan of online job agencies. I prefer LinkedIn. (Student G, PhDEng)

This is again a trial-and-error practice that student jobseekers undergo in their job search process. Many of them have already started to look for best practices in job hunting. Also, since they are still students, their job search is in a passive stage where they are only browsing their options, but they are aware and active in looking for the most effective job search activities.

Job agencies such as Indeed.com send mass emails to a huge number of recipients and when you compare them with LinkedIn, LinkedIn is specified to the jobs for you are looking for, so I can say that that's a real strength when they compare LinkedIn with job agencies, so that's the reason I don't like them. (Student C, M.Eng)

Student G, PhDEng is highly active in looking for the best search activities. He has gone a step beyond and looked for evidence in research:

Comparing this statistic [success rate of LinkedIn] with the number of huge emails I receive from those job agencies, I can say that that's a waste of time, because I have to spend a lot of time reading unrelated job offers.

This quote shows that job searching is an important item on his agenda. Hence, a student jobseeker's search for trials and errors is not limited to looking to peers and following their leads, but if students regard job searching as a significant project, as this student did, they could explore more options on their own terms.

On the other hand, Student I, PhDSc, who is looking for both academic and industry jobs, prefers to keep his options open and explore as many opportunities as possible, stating "I will probably go for as many ways as possible to look for jobs, but I think the most important thing is to look for job sites." Besides, since he refers to job search in the future sense, he is not actively looking for jobs. Also, since he is in his third year of doctoral studies, he is only browsing his options for the time being. It seems that his search for best practices is also passive but he has some objectives, that is, to look for online job search websites. However, he did not indicate that he has actually tried to use the job search engines after he has located them.

As a result, students prefer to use job search websites that cater to specific majors, are equipped with better-customised search options, and are more directly linked with a specific

company's vacancies. Students view companies' profiles on LinkedIn or their Facebook pages. "I always check the social media of job agencies, a couple of related job agencies" (Student A, MSc). They also use occupation-specific online job banks, such as those for the oil and gas industry, like Rigzone.com or Oilandgasjobsearch.com, and academic job banks including Academia.edu or Researchgate.com.

4.2.2.3 Applying directly through the company website. All the students mentioned that applying directly through the company website is an effective way to look and apply for a job. Most of them had a specific company in mind or were familiar with the company either through the career fair held by the career centre or their research projects. They either made a generic profile or applied to a job on the company's website. "I choose the company website. I think that's the best way for us to directly send or target the open positions" (Student F, PhDEng).

All student participants were aware of the companies operating in oil and gas in Canada. They looked for the ones that would match their skills and qualifications. Targeting their potential companies would lead to checking their websites to identify open positions. This practice demonstrates their persistence and interest in seeking employment with their favourite companies. "I always check company websites, you know [*sic*], potential employers, good employers that I wish to get employed by" (Student B, MEng).

Similar to their knowledge of LinkedIn's features, some respondents were aware of different features of company websites. One of the features related to employment that students paid attention to was the ability to follow up on their submitted applications and the functionality of generic applications. Student D, MSc stated he would like to follow up on the applications he submitted to some companies:

Some companies have online form or CV submission link or something, sometimes I've seen some companies don't have anything [*sic*] or they have an HR address or something so you can send your queries or CV to them as well, and sometimes they reply, sometimes they don't, but that's the most direct way to get to the key people ... When you go to the company website, you are in direct contact with the HR.

While some companies do not provide any contact information for further inquiries, some jobseekers might be looking for a more personalized search experience; when they submit their applications, they would like to know if their applications have been viewed by a company's HR. However, regardless of finding a contact, there is still a preference to submit online applications because it is viewed as a direct and targeted strategy in job searching.

Personal experiences of trial and error in search strategies could be significant in future practices in an individual's job search. Once a person identifies a strategy as successful or unsuccessful, chances are likely that they would relay their experiences to their peers, who would perceive such strategies to be models of best or worst practice in the future. For example, one student had submitted an application to a company in another country and the company's HR personnel in his own country contacted him for a job. A jobseeker's chance of finding a job with a multinational company may also be higher by submitting an application through the company's website.

4.2.2.4 Professional associations. All the students were members of at least two professional associations. Once they located a professional association, they would try to find out how the associations could be of use to them. For example, students were familiar with features of such organizations and used them as a source of information either in their job search or finding out about their different events and company affiliations. Taking advantage of job alerts

is one example; once students are members of a professional association, they register their email addresses to receive related or generic job opening emails. Also, students try to track the online presence of associations to learn about their activities and updates because they believe the information provided is valuable.

I registered in a couple of professional associations in engineering such as Professional Engineers and Geoscientists of Newfoundland and Labrador. That's a very valuable information source and also SPE, Society of Petroleum Engineers, so I have registered in the listserv of engineering associations. I'm also checking their social media and their website; I guess that's the best resource for me to learn about the upcoming networking sessions or job fairs. (Student G, PhD.Eng)

An interesting finding is that students found LinkedIn and professional associations to have similar characteristics. More precisely, the focused nature of LinkedIn on specific educational and professional groups demonstrates the same nature of professional associations for some students, that is to say, students realized that both provide specific and targeted information for them.

Professional associations and LinkedIn groups, they have been really helpful for me, because I see that they are really working. I received I can say three or four job positions just from my LinkedIn account, some serious ones. (Student C, MEng)

Data related to students' interest in and use of professional associations show that students are aware of the various benefits of such groups and that their membership is purposeful. Student G, PhDEng identifies the associations to which he belongs:

I'm already a member of a couple of engineering professional associations, such as SPE Society of Petroleum Engineers or ASME American Society of Mechanical Engineers or

NACE, National Association of Corrosion Engineers, those are all the professional associations related to my field of work, so I am already a professional member of all those associations and they all have mentorship programs and they do have some weekly

listserv sending the job openings or positions available related to my field. The student stated that he is "already a professional member of all those associations", which indicates he had a purpose in mind for joining. Students see these groups as a valuable resource and recognize how they can be a beneficial tool in their job search process. One such resource, apart from job alerts, is the mentorship program that these associations offer.

Other offerings of membership include online courses and resume/cover letter writing services. On the other hand, membership to professional associations is not limited to students interested in industry jobs. Interestingly, Student I, PhDSc, who considers choosing academia, research, and teaching, also uses related professional membership to his advantage. Such affiliation gives the students the opportunity to navigate through a wider lens and to learn about employment opportunities and expectations in the academic world.

For academic job search, say for post doc jobs, there are several newsletters or different discussion groups in academia [online], so I have numerous subscriptions to them or membership in different research groups or different associations or societies and they keep posting postdoc positions or even teaching jobs, so I keep getting information from them and I see the job profile, what they are looking for, how many years of experience. So those things keep me informed about what is going on in the field and what they are looking for. (Student I, PhDSc)

4.3 Influencing Factors on Choosing Job Search Strategies

4.3.1 Type of job. A closer look at the data related to the choices of career stream, whether academic or industry, indicates an influence of the type of job on the choice of specific search strategies. For example, in using online job banks, those interested in industry jobs choose to use industry-specific search websites, and those interested in academic positions choose search engines related to academia, research, and teaching. On the other hand, looking for teaching or postdoctoral positions might require the jobseeker to look for other media, such as locating a supervisor who can offer a postdoctoral position or a university with vacant teaching positions. "If it is an academic job, then you'll have to look for a supervisor for a postdoctoral or an institution if it's a teaching job" (Student I, PhDSc).

Networking as a job search strategy is moderated by the type of job student jobseekers are interested in because the providers of such events could have specific affiliations to companies or academia. For example, students who focus on industry jobs look for industry networking opportunities through professional organizations and the career centre, whereas two of the doctoral students (Student I, PhDSc and Student F, PhDEng) referred to conferences as a networking opportunity. Student I prefers a career in academia and Student F's preference is for a position in industrial R&D: "I'm a member of Society of Petroleum Engineering ... I attended their conference ... When I attended the conference I approached the people from the industry" (Student F, PhDEng).

4.3.2 Peers and word of mouth. Reference to success stories of using specific search strategies was evident throughout the data, where the reference was either to a friend or a friend's friend. For example, the most prominent search strategy that students used was networking, since it was reported to be an effective tool for landing a job by friends who had

done so through this method. Adopting the same search strategies as a network of friends is a safe course of action because the friends have already gone through the experience of using a job search strategy, eliminating some of the trial and error for students.

The reason why I refer to networking is because I have heard a lot of stories and experiences like my friends. It hasn't happened to me yet but there is one student who graduated from my lab and was contacted by an employer she had met before. They had talked about her research area. Now she is working in BC ... that's why he stresses that engineers have to network so that's what I was trying even when I attended the conference. (Student B, MEng)

Word of mouth seems to be an influence not only on students' job search choices but also on their aspirations of future career prospects. However, this influence comes more from the trusted networks of friends and contacts. Student F, PhDEng, who was hoping to attain a position in the R&D section of an oil and gas company, was not very hopeful about finding a similar position in Newfoundland; the success story of her friend from another Province motivated her to believe that she might be able to obtain a position elsewhere, regardless of her residential status in Canada. She states, "Not here but I have some friends, for example, they graduated in Ontario and they **did** get a job in the R&D department." Using the word *did* demonstrates reliance and trust in her friends' approaches to job seeking.

The data related to peers' influences and word of mouth is valuable to this study, as it clearly shows that friends are trusted sources of information because their trials and errors provide student jobseekers with reliable information about job search tools. Therefore, students

rely heavily on those resources. The following response is from Student I, PhDSc when asked by the researcher if he had attended the PSDP:⁸

I know a couple of friends who attended the PSDP and the responses were different.

Those who have several years of experience didn't find it too attractive. And on the first day of my PSDP I didn't find it attractive, so I didn't go.

Furthermore, this peer influence applies to choices of other job search methods as well. When it is proved to the student that a specific job search strategy has worked for their friends, they tend to give it a try. An example is submitting online applications to company website:

"I was suggested [to use the online application] by some of my friends. They don't have engineering backgrounds but that's how they got their jobs" (Student B, MEng).

In order to access information on effective job search strategies, students need to build networks of friends. We inquire and hear from our friends about their and other people's job search experiences; circulation of such information among friends could have a further source who is not a friend, but it is important to build information-rich networks. Students look carefully and select the contacts and information that can be of benefit to them. They are also vigilant to the fact that people with the right kind of information are not always in the same major or occupation that they are looking for, but they need to be affiliated to their ultimately preferred sources of information. Once students know their friends have been successful, they trust the method and consider using it.

The right person who is within your network, he or she doesn't have to be the one working in the company but just some information s/he shared like for example, the

⁸ "Professional Skills Development Program (PSDP) for International Students ... helps to prepare international students for professional employment in NL and throughout Canada." Retrieved from http://www.mun.ca/isa/employment/psdp.php

experience I heard from my friends is very encouraging to me because I've heard a lot of stories, finally it comes true when it happens to you, that's why I will keep trying that strategy. (Student D, MSc)

As stated earlier, students as jobseekers look for people to connect them to the right sources of information on career opportunities. These people, as considered by Student I, PhDSc are a "medium" to what they need: "I need the medium, the right people. For example, my supervisor could be the potential candidate for that as well."

The medium that connects the students to job opportunities does not necessarily possess hiring powers. One reason, according to what one of the students had experienced, was whether there are vacant positions available at the time of inquiry. The potentiality of the job therefore could depend on other factors not related to the medium unless the medium is a direct influence, such as the employer or the HR professional.

I would say it really depends on the company and the job opportunity because my supervisor was trying to introduce me to some of his friends in Statoil before, but they were not hiring at that time so they told me to just keep an eye on their website to see if there is any job posting and that's all I got. (Student E, MSc)

In this case, the supervisor connected the student to the employer, but the student was unable to secure a job at the company, either because there were no vacancies in the student's field or because the company specifically followed standard practices of hiring and thus directed the student to their website. On the other hand, the medium could also be social media. In the following example, a friend of Student F, PhDEng had used LinkedIn to contact the company directly for employment information and had been successful in securing a job:

One PhD holder graduate from chemical engineering says he could actually only get the job after he interacted with one HR from one company through LinkedIn and he said other than that he didn't get anything from other people.

It seems that once the student hears about the effectiveness of a job search strategy, s/he would also apply that strategy to his/her search process. The following quote indicates that Student F, PhDEng in this case has used a networking strategy; however, it has not been successful for her *yet*. This might mean that she would continue to use that strategy nonetheless.

Now he [a friend] is in BC. He works in Teck⁹, but that's why he stresses that engineers have to network so that's what I was trying even when I attended the conference. I approached the people from the industry but I haven't got any luck yet.

4.3.3 Type of industry. The nature of the oil and gas industry makes the companies easily stand out because this is a global industry. Top oil and gas companies are known to all students who are in oil and gas-related programs. Therefore, it should not be difficult for them to look up companies on the internet and use the company website to submit general or specific applications. "For those major oil and gas companies, I actually check their website," stated Student A, MSc.

On the other hand, the following example could be an indicator of students' lack of knowledge of the industry's presence in the labour market. It seems that either only larger oil and gas companies are well-known to students, or larger oil and gas companies are among student jobseekers' top choices when it comes to seeking employment. In either case, the popularity of larger companies seems to make them more accessible and visible to student jobseekers, thus

⁹ Tech Resources Limited is a Canadian metals and mining company www.teck.com

they choose to apply to jobs through the company website by either submitting a generic application or applying to specifically advertised job vacancies.

I definitely use the company website because you know they only have a few companies, you can name it, and you can count it, so for larger companies it's a good choice to use their website to find a job. (Student B, MEng)

Students in oil and gas-related programs feel that because their major is limited to the companies in that industry, they have limited options of companies to choose from. This automatically connects them to specific companies, and they do not have to search extensively to find the companies they can work for because they are very much visible.

It actually depends on your major. We don't have many choices. The guy who has a business background can go to any company, or the guy who has accounting knowledge, but for us we have to go to a company which is a geological company doing mining. It is actually quite a narrow field to choose our career. (Student D, MSc)

This limitation also applied to Student F, PhDEng, whose choice was to work in the industry's research and development sector. For example, she mentioned that because not all oil and gas companies have a R&D department, her options are limited. As a result, she already knew which companies she has to focus on:

I want to get into the research and development department, and that's something that only big oil and gas companies have. So for example, my priority is Shell but there is Statoil and other companies I'd like to go to as well ... Shell is kind of [*sic*], invests a lot of effort or they have a lot of research and development facility, I mean it's pretty good on the R&D part.

Students' trials and errors in their job searching go beyond job search strategies and could include any opportunities that would give job-related information to the student jobseeker, such as cooperative education. On the other hand, collaboration between the industry and university could eventually provide such opportunities. The following example demonstrates how Student F's experience with cooperative education has given her an opportunity to gain a better insight into her chances of employment with the company:

I did co-op. It didn't work out for me because Husky doesn't have R&D here and when I was doing my internship here, they mentioned that because I have engineering research background [they are going to offer me an internship as a project related to my major] and they actually have a small project going on, that's why they collaborate with university and that's why I got that position but other than that they don't really look for

PhD holders here. It's kind of frustrating but that's what I learned from my experience. Her feelings of frustration due to the insecurity of her future employment chances with Husky, as she mentioned, could indicate that she would be more selective in her future job hunting endeavours. This proves that the type of industry can affect students' choices of job search strategies in oil and gas because there are different departments in various companies that require different skill sets. As a result, students might, for example, use the internet the most frequently to locate companies that provide them with specific employment options.

4.3.4 Having prior work experience. The data shows that graduate students with previous industry work experience tend to use identified and specified search strategies, including job alerts from professional associations and connecting with potential HR professionals or employers in the industry through networking and LinkedIn. These search methods are direct and targeted and could connect students directly to job leads. In their

statements regarding the use of job search strategies, students used the words *effective* and *successful*, and all of them seemed knowledgeable of the job search process when they referred to the different steps it involves. Their knowledge was a result of their experience using search strategies that secured them their previous jobs and hence were successful and effective, whereas students without previous related work experience relied mainly on university networking sessions and online job banks.

Both experienced and inexperienced students perceived applying directly to companies' websites as the most effective way to reach employers directly. Hence, these students were also looking to find best practices of job search. On the other hand, two students with previous industry experience (Student I, PhDSc and Student E, MSc) did not find PSDP effective in meeting their job search needs, since they had already worked in multinational companies and were aware of an international work culture; in addition, they had exercised career preparation and learned the best practices in job searching. These students were interested to learn more about tools that are targeted at connecting them with job openings directly. They believed that the program was more suitable for entry-level students to help them learn about the basics of job searching.

I went to one class of PSDP two years back and I think it wasn't the best thing for me. I think it's good if you are looking for a job for the first time but if you have 3-4 years of experience in the industry, it is not for you. (Student I, PhDSc)

In another instance, a student without previous work experience referred to volunteer work opportunities as networking:

There was a volunteer fair in the landing.¹⁰ 'Cause [*sic*] I don't have volunteer experience. It's another factor that affects you to find a job. Through that I joined the volunteer job. So it's another event I joined. (Student D, MSc)

First, Student D perceives these opportunities as factors that would influence his job search efforts by networking with local people and improving his communication skills. Second, he feels that he can gain some experience through volunteer work due to his lack of work experience; thus, the volunteer work could lead him to connect with new people and help him build skills in the meantime.

4.3.5 University career centre (UCC). A more comprehensive look at the data from the interviews reveals a positive pattern of influence from the UCC and its services in students' job search processes, as well as their choice in using job search strategies. All the students attended the resume and cover letter workshops at least once, job fairs, employer information sessions, and other networking events held by the centre, and generally scheduled appointments for career counselling.

Whenever I want someone to review my cover letter, CV or resume, I go to the career centre, so I can say that I use their resume and cover letter service at least twice per term. I was looking for a part-time job here last summer; they were really helpful for me to find a job. I just went there and talked to them about the type of job I preferred to fill my part time and they offered, you know, they proposed a couple of options and that was really helpful for me. (Student G, PhDEng)

Resume and cover letter writing services are used mostly by international students, which is understandable since they come from different work cultures and would like to improve their

¹⁰ Landing is the name of a student gathering space on campus.

chances of employment success. The service provides them with essential information about how to write a resume and cover letter and what to include. Students recognize this as an important service because they are aware of the need to adapt to cultural differences.

I went there to show them my CV and ask them, if I want to apply for some part time job here, say Aquarena¹¹, then this is my CV, how can I orient it to look for a job here. Because that CV was built 5 years back and it was for Indian employers, so how would I modify that? (Student I, PhDEng)

The UCC was known to all the participants and all of them perceived it as effective and helpful in their job search process. Student I, in his third year of doctoral studies, showed interest in using the UCC's services in his final year or after graduation (i.e., during his active job search period): "Regarding the services that they have such as mock interview or resume and cover letter review, those services are really helpful, so I will also use those services when I start looking for a professional job in the future."

Data related to the UCC sheds light upon its influence on students' choices of job search strategies. By offering various services and networking events that target career preparation, the UCC has a great influence on the students' available choices. One of the programs they offer exclusively for international students is PSDP. This is a career preparation program that familiarizes students with the Canadian labour market culture and ends with a networking session. One of the students' comments upon attending this program affirms its usefulness: "As a part of PSDP program, they had one of the networking sessions and that really worked for me, I really found a job. (Student C, MEng)

¹¹ Aquarena is the university's swimming pool.

The UCC offers exclusive sessions on job search strategies and how and where to locate job vacancies within the Province. Although none of the students were aware of this service, they were exposed to some information on job search strategies through PSDP. On one hand, there is an exclusive program targeting one specific area in the job search process; on the other hand, there is a more inclusive program that takes a longer time to complete and requires a greater commitment compared to the one-hour information session on job search strategies. The inclusive nature of PSDP, with each session on a different topic of job search, provides students with a job-related information package.

I can say 100% yeah [*sic*], because I didn't use my LinkedIn that much actively before attending PSDP and when they provided a couple of statistics comparing how employers use those social media to find the employees, I can say that I renovated my LinkedIn profile and changed lots of information, yeah [*sic*] they definitely have enhanced my job search process. (Student G, PhDEng)

This clearly shows that the UCC has had a direct influence on the student's choice of job search strategy (i.e., LinkedIn), apart from networking.

4.4 Emergent Themes

The aim of this section is to look at the data beyond the initial research questions as a result of further in-depth analysis of the interviews. The following themes were determined to be recurring by reading the transcriptions more carefully, paying unbiased attention to the feelings and tone of the speakers, as well as taking the attitudes of the students into careful consideration.

4.4.1 Job searching as a planned project. It is interesting that all the students talked about their job search process with confidence. They all answered questions without hesitation and clearly showed that their choice of using job search methods was purposeful. The students

acquired this confidence by observing their peers and evaluating the peers' successes or failures in using job search strategies. Their eagerness in answering the researcher's questions was evident in their use of powerful words that exuded confidence. The following quote from Student A, MSc illustrates a prepared mentality towards the job search process:

You know, **based on my experience** from my graduated friends, finding a job is not easy. You **should** look at it as <u>a real project</u>; you **should** <u>specify</u> it, big chunk [*sic*] of time for <u>this project</u>, so time is **really important**, and you **should** start <u>ahead of time</u> before graduation **at least** to <u>identify</u> your <u>objectives</u>.

The bolded words emphasize the intensity of emotion conveyed. For example, the phrase "based on my experience" indicates the respondent's confidence in his job search experience, although he refers to his friend's experience as if it was his own. There is a sense of trust that connects experiences; therefore, as his friends had difficulty finding a job, he also perceives job searching in the same way. Additionally, multiple uses of the word "should" and using emphasis (e.g., "really important" and "at least") reveal how the respondent feels a need to evaluate job search as if he was measuring the importance of job search planning on a scale. He refers to job search as a "real project" that requires proper planning in advance of graduation. This could mean that before graduation, students' career planning is in a preparatory stage, which needs specification of time and identification of objectives ahead of graduation in order to prepare them for active job search upon graduation. Hence, students were clear about their objectives and assessed their situations carefully.

You **should** be <u>clear with yourself</u> which <u>location</u> you <u>are willing</u> to work, which <u>locations</u> you really <u>don't want to</u> work, or what <u>companies before graduation</u>. I **should** be <u>clear with myself</u> what my <u>objectives</u> are, I **should** try different options through

<u>different information resources</u> and <u>different networking sessions</u> to <u>clarify my objectives</u> to myself, and that **needs** <u>time</u>. That is **really important** <u>before graduation</u>, and <u>after</u> <u>graduation</u> I can say that the <u>real project</u> **starts**, so you **have to** <u>specify</u> to look. You know, job search **in my opinion** is <u>a full time job itself</u> (chuckles), so <u>after graduation</u> I can say that you **should** spend <u>your full time</u> to <u>not lose available opportunity</u>, <u>which you</u> <u>have at the time of your graduation</u>, so I can say that <u>the time</u> is **really important** both <u>before graduation</u> and <u>after graduation</u>. (Student G, PhDEng)

This quote is another example of the awareness of the demands involved in a job search process, which involves clear objectives, time, and planning. The student is confident in his response, and similar to the remarks made by the previous student, this student uses many words (in bolded font) that indicate his assertiveness in the process. He reached a sudden realization and conclusion by his reflections on job search and then chuckled when he referred to it as a "full time project." His reflections led to multiple references to the "time" associated with graduation. He advises that before graduation students need to find out what they want in a career. For example, they need to identify what companies and locations they prefer to work in and shortlist their priorities, take advantage of available resources, and reach out to find even more resources, such as "different information resources" and "different networking sessions," so that after graduation, they would have the necessary resources based on their objectives.

The job search process in the preparatory stage before graduation was purposeful and planned for all respondents. Planning involves a careful search and selection of desired jobs for reference in the future, potentially to be used during active job searching. The following quote shows the respondent's confidence in job searching by demonstrating an objective for the choices of career and knowledge in the job search process:

I actually started to do this long back. Before even I saw [*sic*] one position show up, I would actually put the information in one file I kept for the companies in record ... turns out by now I have quite a list in my hand. (Student F, PhDEng)

Students' expressions of their career and job search plans suggested confidence and an understanding of a need to be vigilant towards their search objectives. The description of another student's (Student I, PhDSc) job search process shows careful reflection on his career aims and aspirations. From his previous industry work experience he realized that a position in industry did not interest him, thus he came to Canada to pursue doctoral studies in order to change his career path and gain a position in academia, teaching, and research. In university, he was given the chance to experience teaching and research, which became his career aspiration. After his industry experience, his decision to gain initial experience in the academic world was thoughtfully planned, which eventually widened his career options: "Both **Plans A and B** are two different thoughts because these two are <u>different clear thoughts</u>: one is more going into the academia, more like postdoc research and all and another one is <u>coming back</u> to industry."

Having an objective in job searching was evident from the respondents' sense of responsibility for their future career, which led to identifying their skills gap and finding ways to enhance or gain required skills. On the other hand, they also realized that they had to start this preparation early to have sufficient time to locate the resources they needed for skills enhancement.

If I go for academia, which is my **first preference** ... I did a course in education department which is **Teaching Skills Enhancement program (TSEP)**, so I did that, completed it <u>successfully</u>, learned a lot and they have <u>an awesome thing</u> called teaching dossier preparation, so teaching dossier is **the first thing you need** if you want to apply

for any teaching job whether now or after two-four years after having <u>some start-up</u> <u>experience</u>... and teaching dossier is something that <u>you build over time</u>, it's not something that you build in one day and finish. (Student I, PhDSc)

The awareness of time and readiness in job searching was not exclusive to students with previous experience and applied to those without work experience as well. The following viewpoint is expressed by Student D, a master's student without previous experience:

The oil companies have **specific jobs for new graduates**; actually, it's called **the new grad program**. It's more like <u>a shift</u>. Ok, you do this position <u>for a couple of months</u> and then you <u>change to another position</u> because you are <u>circulating to see</u> how jobs are working. <u>There are certain jobs for new grads</u>.

This shows that having work experience did not seem to determine the student's awareness of his career objective in the search processes. The response is a good example of the student's proactiveness in searching and identifying his career options as a new graduate in the future, as well as identifying resources that could enhance his chances of success. Awareness of new graduate programs of oil companies and others would help students determine how they need to prepare themselves for the requirements of the program. The student's last statement, "There are certain jobs for new grads," indicates that he has searched and located what companies can offer specifically to new graduates.

Student I, PhDSc specifically mentioned that he considered his school years to be a time for active job search preparation. By identifying his current situation as one that allows him to construct purposeful skills or tools for the real job search upon graduation, he reveals that his whole vision of the job search process is determined by a measurement of time. The time to study is the preparation stage and the time to graduate necessitates active job searching using the

tools and resources that he has prepared for this stage. His multiple references to reasoning by using words such as *so* and *unless* demonstrate how critically he navigates the process of job searching:

I think <u>at this stage</u> I am **more focused** on doing my research because whatever I produce in terms of papers or publications, or whatever I do in terms of analytical experiences, those <u>will be counted</u> **when** I start looking for a job in academia or even in the industry <u>so</u> now I am **more focused** on doing those things than looking for a job ... so **at the moment** I'm **not giving a lot of** time to job search because that doesn't mean anything to me <u>unless</u> I have my resume or teaching dossier or whatever tools I <u>have built</u> or I am <u>in</u> <u>the process of building</u>.

Touching upon the notion of time measurement in students' responses to job search preparation is a quote from Student E, MSc that contains many action verbs:

Once, say, I **will be** ready for job search, then I think I'll have **a schedule** of my **priorities** and what to look for. **If** there are, say, a certain number of job advertisements which are suitable for my profile then I'll **short list** them, **save** them somewhere, and then I'll **go through** <u>each</u> of them and I **orient** my CV, which needs a lot of work ... I think you **should look beyond** job advertisement, a little bit about the person or the organization to **have a feel of** what is the <u>philosophy</u>, what is the <u>motto</u>, **reflect that** on your CV and then the cover letter. I think the people who are sitting there to select the

CVs, they can pick yours saying this guy overall has <u>more matching criteria</u> with us. Once again, the sense of responsibility in having an objective in searching originates from the student's voice, using such action verbs as *short list, save, go through, orient, look beyond*, and *reflect*. Then, the job search process is presented through visualizing the stages by using time and

conditional clauses; therefore, one action leads to another. For example, the phrase "Once, say, I will be ready for job search" refers to the active job search period after the preparatory search phase. The student has the knowledge of what is required of him during the active job search, such as having practical information about the company he would like to apply for (i.e., to go a step ahead of regular job searching and application by trying to understand the company's culture, philosophy, and motto), and then to incorporate those elements into his application. Thus, is a good example of strategic job search planning by the student.

4.4.2 Students' perceptions of the oil and gas labour market of the Province. The knowledge of the Canadian labour market related to job search and recruitment strategies is that networking and referrals are the most important gateways for the students to connect with HR professionals and eventually land jobs. Having social contacts was frequently referred to by students as the most effective job search strategy in Canada: "The **friends** who work at the company can **recommend** us to go to the company" (Student A, MSc). This perception came from learning about their friends' successes in job finding through networking and contacts, which led to counting such friends as potential contacts that could result in jobs. Networking and referrals have been successful for students in landing jobs, and hence it is seen as the best job search practice by international students. However, it comes with criticism. The following quote shows that the student is aware of the popularity of networking and referral systems in the Canadian labour market:

I would say, and overall **in Canada**, I think the general trend in the industry is very **recommendation-based**, which **is good to some extent** but I'm **not a big believer** in just recommendation system which **is very famous here.** (Student C, MEng)

He refers to the system as "recommendation-based" and he is does not subscribe to it. Further discussion with him clarified that by saying that he was "not a big believer," he meant that he did not believe that this system would be efficient for the whole population of jobseekers; if companies relied solely on this method, they simply may not access a wider pool of applicants who might also possess the right talent.

Most of the students had a shortlist of preferred companies, and Student D, MSc specifically mentioned that his chances of employment with his favourite company depended on the company's hiring discretion with regard to his lack of industry experience. Regardless, he was still determined to apply and had also planned to identify and apply to smaller companies in case his first attempt failed:

It **depends** on the company. It's **not my decision**. The **company chooses** us ... I **targeted** at the Canadian companies like Suncor, **the largest** Canadian company in Canada for oil and gas, which is **one of my first choices**, but if they turn me down, I can **apply** to smaller companies at first.

This quote expresses the student's perceptions of the larger oil and gas corporations having streamlined hiring behaviours that make it harder for entry-level graduates to "get their feet in the door"; despite his confidence, his views might encourage him to apply to smaller companies. On the other hand, Student E (MSc), who has prior industry experience, sounded eager to stay in Canada after graduation and was confident that he would be given a job offer, believing that the job market in Canada is not competitive for him and accepting an offer would depend on his own decision:

I **would love to stay** here but it **depends** on the offer. If the offer is <u>good enough</u>, good in terms of say <u>if I love the job</u>, if there is <u>good growth potential</u>, all these things, so then I would love to stay.

In other words, he would assess his future job satisfaction with the criteria he considers for his job offers (i.e., a job that would provide him with satisfaction in terms of professional development).

The fact that the oil and gas industry is not limited to Newfoundland provides jobseekers with some freedom in terms of their location of employment; however, diversity plays an important role in their perceptions of oil and gas companies. Based on the attitudes of the students about the Canadian and local employers and companies, it seems that they highly value diversity as a potentially significant condition for their future workplace, and hence they assess the Canadian job market in terms of multiculturalism. Therefore, students perceive that lack of diversity is a barrier to their job search in Newfoundland, and as a result, they are open to consider their options in other parts of the country. One example of this belief is described as follows:

We know that Calgary or **Alberta** has a **great potential** for oil field and if you go to say Saskatchewan or Quebec or if you **go to** British Columbia, there are **lots of mining activities** happen there now [*sic*], so **if** I want to focus on a city, I **would go** to those cities where there are **more international people** and work market is more **diverse**. I think in **Newfoundland** the **industry** and the **mindset** is still not that open so it's **more difficult** to get a job here, so I **would look for** those places in **Canada**. (Student B, MEng)

Employment for international students in a foreign country has its own restrictions and some companies may base their hiring criteria on the residency status of candidates. Likewise, students showed some concern about their chances of employment after graduation, while they are still a foreign national. Although most of the oil and gas companies are multinational, it appears that some, nonetheless, prefer candidates with stable residency status; hence, students perceive it as a barrier to their search process:

For oil and gas companies for example, ExxonMobil or Imperial Oil, they actually **prefer** people who **already have citizenship or PR¹² status**. For me, I don't have any of them so for me, maybe it's not, uuhh [*sic*] what to say, the only **restriction** for me but to me it's a kind of **barrier**. For me when I'm **looking for** a job and if the company says, ok we **only prefer PR or Canadian citizens**, I **probably** skip the company, because I **tried** and it **didn't turn out** good. I **sent out** my resume to the companies, **maybe** that's not the only reason but I **feel I didn't hear** anything but Shell is kind of [*sic*] **open** in that part. (Student E, MSc)

Students' perceptions of diversity in Newfoundland's labour market are a result of their personal experiences, but these are also based on what their friends and people in their social networks have experienced. Their friends' perceptions and experiences of labour market characteristics, such as diversity, can affect students' views and thinking as well:

I haven't worked here but I say that **based on** my **friends and people who are working here** or who are looking for jobs here, according to them the job market here, **if** you go to **different companies**, and if you look at the **employees**, it's **not very diverse**, so that tells you something about it, **right?** (Student C, MEng)

¹² Permanent Residency

Despite constant comparison of their eligibility against companies' standards in terms of residency status, students are motivated by other available options in terms of workplace locations in other provinces:

It's not that all the **international students** who finish their degree here are not **eligible** to work, or are not **good enough** candidates, because all of these people are **going out** of the province and working **somewhere else**, doing jobs, and doing **fine** and other provinces are doing I think good, even **better than Newfoundland**, so there is something wrong at least in the **mindset** or in the **perspective** of the **employers**. That's why the job market is **not very diverse** and open to **diverse communities**. **Yeah**, that's what **my feeling** is. (Student I, PhDSc)

Student G (PhDEng), who has been at Memorial University for more than three years and has quite a large network of friends, comments on the trend he has witnessed on international students' employment in the Province: "Every year, most of the students from MUN are **not staying** here; they are **leaving**. Only a **very small percentage** of students are **getting jobs**." His comments show that he, like other international students, is evaluating his options to stay and work in the Province, but he is also making an effort to understand what his and his friends' barriers are in terms of finding employment opportunities.

4.4.3 Recommendations from Students

Students responded to questions pertaining to on-campus career resources. They were asked about the resources and services required to connect them to job search strategies and to provide recommendations for the UCC. Their responses denoted they would like to be provided with more discipline-specific networking sessions throughout the academic semesters. The recommendations also apply to job fairs because students believed that most oil and gas

companies aimed to recruit undergraduate co-operative students at these events, so graduate students were only able to network and ask industry- and job-related questions. They also suggested that employers go beyond targeting the recruitment of co-operative students at job fairs and recruit graduate students directly on campus, where active interview booths could be set up. For example, Student E recommended that the UCC could host technical fairs where graduate students would be able to showcase their research projects to potential employers. Another idea, suggested by Student I, was to have a two-level PSDP program targeting students with prior work experience. Here is a direct quote from Student I, PhDSc about needed improvements to PSDP:

I think PSDP needs to be more specialized and improved for those people who have certain job experience. Because when you go to an employer with four years of job experience, then the type of interview questions that you face will be different for you compared to the first time jobseekers with no experience because interviewers change their questions according to what you have. I think PSDP is good for basic training but those who have worked several years and have had several interviews and prepared already in some way, for them PSDP needs to be more advanced.

Student F, PhDEng, who was looking for R&D positions, wondered where she could learn more about companies that have R&D sectors, apart from checking company websites one by one and attending networking sessions. Student D, MSc suggested having programs where employers are invited as mentors to speak to students in depth about the skills they value in an applicant.

Chapter 5

Discussion

5.1 Introduction

The Discussion is based on the results of the study described in Chapter 4. The major findings of the study were synthesized with reference to Chapter 2, which provides background knowledge from the literature for this study. Therefore, it is essential to understand how the literature review provides insight in terms of a discussion of comparisons, contrasts, evidence, and explanations based on the findings of this study. The first major finding is that students perceived networking and online job searching as the most effective job search strategies, and thus they used these the most often; second, certain factors influenced their choices of using such job search strategies, which included a) the type of job sought, b) the influence of peers and word of mouth, c) the type of industry the individual is interested, d) whether or not the person had previous work experience, and e) if the individual had assistance from the campus career centre.

The other findings were based on two emergent themes: a) job search as a planned project and b) students' perceptions of the oil and gas labour market of the Province. In this chapter, the results are presented in two sections. First, major results and influencing factors are provided in light of the finding that students' job search processes were planned with demonstrated goal setting in using search strategies; the second section contributes to the discussion of students' perceptions of the Province's oil and gas labour market.

5.2 International Students' Job Search Process

International students as research participants in this study included those with and without previous work experience and who indicated an interest in seeking employment in

Canada upon graduation. Students with previous work experience had oil and gas-related jobs in industry, while those who were entry-level applicants did not report any prior work experience.

Students are in university to study, but they also use their academic time as a preparatory process for a more thorough career search upon graduation. The purposefulness of this process was apparent throughout their responses in interviews by reflecting upon and visualizing the search process. Multiple uses of future tense and conditional sentences, and references to time in their job search processes, demonstrated a feeling of responsibility for their future careers. In addition, distinguishing between the time before and after graduation clearly indicated how well they had thought through the process. Job searching for international student jobseekers in this study was a strategic plan that consisted of prioritization, knowledge of potential companies beyond job descriptions, and knowledge of one's expectations in job search and career choice. This whole presentation of thoughts, feelings, and actions towards job searching indicated that students had a high level of confidence in their preparatory search processes and job prospects.

All participants, regardless of having prior industry work experience, were knowledgeable of the oil and gas labour market and were aware of their career options; however, students with previous work experience had better job search clarity (Wanberg, Hough, & Song, 2002), that is, they had prepared a shortlist of their desired companies, location, and job titles. Their awareness and knowledge resulted from their trials and errors in attempt to find the best job search strategies, in addition to industry-related information gained through talking to friends and acquaintances, attending networking events, browsing company profiles on the web, and using the university career centre.

Students' references to the uses of job search strategies were in the form of narrating a visualized process by reflecting on their job search and career objectives. The reflections

indicated that they were aware of the available time during their studies and considered it a period of "job search preparation." During this period, student jobseekers sought information on job search strategies and potential companies in order to be ready for their job search upon graduation. Before graduation, their search attempts were preparatory, because despite the possibility of applying to some jobs, their actions were only a means of trial and error in job search methods. By using different search strategies, some students mainly searched for job openings to learn about job descriptions and their requirements. This period also included gathering information and enhancing their skills. Kimmell (2003) refers to this method of job search as "passive search strategy," since the jobseeker only looks for job vacancies using different strategies, whereas applying to job openings after locating them by using job search methods is called "active job search strategy" (Royal, 2001). On the other hand, Boswell, Zimmerman, and Swider (2012) provide a different definition of active job search. They define it as any job information-seeking strategy that involves communicating with social contacts. Considering all three definitions of search strategies, students' preparatory stage of job search consisted of both passive and active job search strategies.

Knowledge sharing was the major component of the participants' information seeking during the preparation period. This occurred through friends who had already undergone trials and errors in using various job search strategies, so their experiences were regarded as reliable measures of effectiveness. Thus, one of the most influential factors found in this study is the impact of friends and network of peers and their friends on the students' choices of job search strategies. This factor was viewed as one built upon trust, trial and error, and assessment of the sources of information circulated by word of mouth within their network. Word of mouth has positive effects on job search outcomes (Van Hoye, 2013), and students have realized this by

hearing stories of successful outcomes from their own friends and friends' peers. Recommended search strategies were considered effective because they had either yielded jobs or provided better job information for the candidates. Therefore, students attempted to seek information-rich networks through online and face-to-face networking. In order to obtain positive results, they also needed to exchange and be the carriers of information. As a result, networking alone is insufficient and students must transfer valuable information to employers by talking face-to-face (Sanyal & Neves, 1998), as the students in this study attempted to discuss their projects with employers. Thus, word of mouth suggests that networking is a successful search strategy.

Sanyal and Neves (1998) described networking as a process with three stages: a) planning, b) connecting (i.e., making contact), and c) sharing information. Students' reflections on using networking as a job search strategy resulted in providing a definition that closely matched this definition; students' also generated some characteristics for networking based on their own interpretations and experiences. The discussion to follow is a comparison between students' and Sanyal and Neves's definitions of networking, followed by a discussion on the characteristics of networking presented through students' interpretations.

Students perceived job search as a "real project" that would require planning with objectives and strategizing. Similarly, networking is a job search process rather than a one-timeuse method that demands strategic planning and goal setting. For example, having clear objectives before executing a job search strategy, such as networking, helped students to recognize how and where this strategy was beneficial in their job search. One of the objectives of networking for them was to find potential contacts, i.e., people who could act as catalysts to connect them to job leads. Such people could be viewed as carriers of information on job openings as well. Another objective was to improve soft skills, which was demonstrated and

emphasized by a student without prior work experience. Furthermore, in order to plan for accessing this search strategy, students were vigilant and proactive. They received email notifications from the university career centre on the various networking sessions happening on campus and joined professional associations to learn about industry and community networking opportunities. In addition, they often attended these networking sessions. Finally, their planning was strategic in that they narrowed down their lists of social contacts based on people's professions and knowledge related to oil and gas industry jobs.

This strategic plan or real project starts with "planning," as proposed by Sanyal and Neves (1998). This stage involves objectives for using and strategies for accessing and selecting resources, such as networking opportunities offered by the career centre or professional associations that further result in connecting student jobseekers with potential contacts. For instance, international graduate students reported that their purpose of going to job fairs was to gain job-related information by talking directly to employers. For the jobseekers to be successful, it is important to have face-to-face conversations with members of industry and keep updated on the changes in their field (Scheele & Madison, 2010).

On campus, student jobseekers learned about networking opportunities either through the career centre or by their friends' recommendations to attend them. For instance, some of the benefits of job fairs that led to such recommendations were that students were able to exchange research- and industry-related information with employers or HR professionals at the event. Students were aware that talking directly to employers would enable them to gain inside knowledge about the companies and their projects and learn beyond job descriptions.

In their accounts of the networking process, students referenced many different variables of networking that are described here as characteristics. Using networking as a strategy to seek

employment means that the jobseeker may have to allocate some time to plan his/her strategies and objectives prior to commencing a serious and active job search period, which in this case would be prior to graduation. Networking and building contacts, and finally selecting potential contacts, could be a lengthy process, and hence time is an important factor in networking. On the other hand, networking is viewed as a tool to help student jobseekers gain quality job-related information. Once social contacts are established, they are potential sources of and a means to access job-related information. Thus, people are vehicles of information, and along with knowledge sharing, are important components of networking. In other words, the purpose of selecting potential contacts is to find people who can lead or connect jobseekers to job opportunities. Students were mindful of this purpose, and thus they made attempts to gain access to such information-rich individuals by attending networking sessions, talking directly to employers, and sharing knowledge. In this study, the desired contacts were employed in the oil and gas industry or had contacts with people in the industry; accessing these individuals was one of the objectives of networking that influenced students to use this method in seeking employment. The goal to seek information on the Canadian labour market would give the student jobseekers an opportunity to place themselves "on top of the competition" in the job market. Students valued opportunities to talk face-to-face with employers to discuss their own research projects, and as one student said, "to get their face out there."

Networking events such as job fairs, employer information sessions, PSDP and ETP networking sessions,¹³ speed networking, industry networking sessions advertised by professional associations, students' research project presentations for the sponsor companies, and volunteer opportunities, could be considered strategies to meet and communicate with people in

¹³ Each of these programs ends with a networking session between the student participants and community leaders and employers.

order to share information. Moreover, networking was perceived as a preparatory job search strategy that requires time and planning, since building a network of contacts can be a lengthy process; student jobseekers might need to maintain and build contacts over time in order to be able to access them at the time of active job search (i.e., upon graduation).

Thomas (1989) divided job search behaviours into three categories: traditional, nontraditional, and networking. Traditional strategies involve searching via newspaper advertisements, job listings and/or employer interviews at placement, and off-campus employment agencies. Non-traditional strategies include mailing resumes to employers and calling or visiting employers directly. Networking is the act of establishing contacts, which may include faculty, academic advisors, alumni, family, friends, and/or placement services staff. In his study, business, science, and engineering students used non-traditional search strategies more than traditional methods and networking. Thomas' findings on students' preference to approach employers directly were consistent with the findings in this study, as the participants preferred direct approaches such as applying to jobs on a company website and talking directly to employers during networking events. However, students' respective references to the notion of "networking" suggest different connotations of the word. "Networking," according to the participants of this study, is "any form of face-to-face or virtual contact with the aim of getting job leads," which could be random or purposeful. As a result, what Thomas (1989) considers a non-traditional search strategy (i.e., "visiting the employer directly") falls into the networking category based on the participants' views. Therefore, there is a possibility that researchers and the participants might categorize job search strategies differently.

The importance of networking to international students in this study could be attributed to an initial absence of personal or social contacts in Canada or the city in which they will reside.

Building social capital for international students in a new place requires some planned efforts in their short period of study. As previously discussed, they made planned and purposeful attempts to build contacts. Employers and other people that students meet during networking events are those who do not have regular contact with students and are called "weak ties" (Yamkovenko & Hatala, 2014). Students often consider networking through weak ties to be an essential part of their job search process (Marmaros & Sacerdote, 2002). Social networks provide us with information (Kim, 1994); thus, students as potential jobseekers have to build new social networks to be able to access information on jobs. On the other hand, informal job searches, such as approaching "strong ties" (e.g., friends, relatives, and people who know the jobseeker; Cahuc and Fontaine, 2009), could generate more leads and improved access to job descriptions, making this one of the most effective means (OECD, 1999). Notwithstanding, students try to build weak and strong ties early, despite the difficulty of creating strong ties in such a short academic lifetime (White, Boorman, & Breiger, 1976), as they have very few or no contacts when they first arrive to study in Canada.

Students in this study tried to attend as many networking sessions as possible to expand their social influence and then identify and select the ones that could lead them to potential career opportunities. The actions of these students reflect the findings of Granovetter (1995), and Marmaros and Sacerdote (2002), who discovered that weak ties were the students' preferred sources in searching for jobs. However, Yamkovenko and Hatala (2014) believed that weak ties do not provide us with detailed information about jobs. This is contrary to the findings of this study, because one of the reasons students preferred networking and building weak ties was to find out information about companies, job openings, ongoing projects, and more importantly, the skills that such companies were looking for in an applicant.

Following the students' interpretations, one of the most important outcomes of using networking to seek employment is to build a substantial network of social contacts; therefore, it is necessary to start early so that student jobseekers have enough time to do so. Another networking strategy for employment purposes was "referrals," that is, when employees refer a person to their company's HR professional for employment. International students were aware that referrals are reported to be one of the most preferred ways of recruitment by oil and gas recruitment professionals (Ernst & Young, 2011). Consequently, they made efforts to build networks of weak and strong ties in order to be more accessible in the job market. To increase their chances of employment, they exhibited an eagerness to learn about the skills and qualifications that employers were seeking in applicants. They attended the networking sessions not only to make weak ties, but most importantly to learn about their work culture and environment.

As previously stated, the aim of international graduate students' visits to job fairs on campus was to gain industry- and job-related information. This differs from the aim of undergraduate students: to discuss co-operative education placements with employers. The Department of Advanced Education and Skills of the NL Government (2007) reported recommendations addressing higher education institutes in the Province to "expand co-op programs and other forms of experiential learning that encourage attachment to the workplace" (p. 60). Co-operative education and internships are mandatory components of undergraduate engineering courses, whereas graduate students are rarely able to take advantage of such opportunities. Considering the job search behaviour of the students, it is noteworthy that students in co-op programs demonstrate greater confidence and realism towards their career objectives, and they are more engaged with campus career services (Ng & Burke, 2006). Since the

participants (international graduate students) did not have a co-op or internship component to their programs, it was uncommon for them to receive placements with companies unless the coop or internship position was initiated by a contact based on industry need, a scenario experienced by a master's and doctoral student in this study.

Another search strategy that international graduate students preferred was the internet. In previous studies, post-secondary students most frequently chose the internet as a job search tool because of its convenience and cost effectiveness (Lin, 2010; McKeown & Lindorff, 2011; Wanberg, et al., 2000); however, in this study, graduate students preferred the internet because it provided them with access to various methods of online job searching and networking. LinkedIn, a professional social networking website (Sułkowski & Chmielecki, 2014; Bissola & Imperatori, 2013), was the most preferred online tool for job searching and networking while being the mostused social medium for recruitment by employers (Bissola & Imperatori, 2013). Herbold and Douma (2013) found that graduate students used online social networking more than students at other levels of education. Similarly, graduate students in this study preferred LinkedIn as a networking tool and used it in the preliminary stage of the job search process (i.e., before graduation). However, they used LinkedIn for other purposes related to job search, such as joining professional groups related to their field of study and following company profiles. In addition, they used it as a tool to connect to HR professionals, employers, and other people in the oil and gas industry. While Sin and Kim (2013) found that international students use social networking sites as a means to access information rather than building social capital and networking, in this study, students used LinkedIn for both purposes: to access job-related information and to build social capital. On the other hand, LinkedIn as a networking tool (Bissola & Imperatori, 2013; Sułkowski & Chmielecki, 2014) provides an opportunity for

students to connect with those beyond their own networks (Hempel, 2010; Karlin, 2005). This feature is recognized by the students in this study to be particularly valuable, as they not only look for contacts in their own field of study and work, but also for people who can connect them to such contacts. Therefore, the wealth of information such networks provide could lead to potential jobs; as the number of interactions increase, students access higher degrees of social contact (Kim, 1994).

Networking to build contacts can be a lengthy process, but perhaps what makes social media popular among professionals and jobseekers is that it can shorten the time required to build meaningful contacts (McKeown & Lindorff, 2011), for example, by using the search engine on LinkedIn. This feature alone gives jobseekers exposure to potential contacts such as HR professionals; once a person visits a profile, the owner of the profile is notified and can browse the jobseeker's profile in return, increasing the chances of the jobseeker's employment (Garge & Telang, 2011).

It is important for students to know and learn about the various functions and uses of LinkedIn. For example, the job search engine on LinkedIn provides targeted information, i.e., jobseekers can search beyond locations and job titles; the search is extended to companies and even to find people with specific job titles, locations, and workplaces. However, jobseekers need to keep their profiles updated in order to maximize their chances of employment (Scheele & Madison, 2010).

Royal (2001) suggested that jobseekers learn about the companies before approaching them for employment. Students in this study sought any opportunities for face-to-face encounters with employers, including career fairs, employer information sessions, and other networking sessions to inquire about their various ongoing projects before sending their applications through

the company websites. However, once they encountered the employers, they were often directed to the company website. On these websites, students were able to read about the company and job descriptions, and they could apply to jobs or submit a generic application to their preferred companies; consequently, applying directly to the company via website was another method used by all the students. This method was chosen because students believed that this is another way to approach employers directly (Bissola & Imperatori, 2013), which could increase the likelihood of being hired (Kuhn & Skuterud, 2002).

Students preferred any kind of search activity that is focused, targeted, and directs them to job openings. Online job searching provides streamlined and customized search engines that are directly related and linked to the oil and gas industry (e.g., Rigzone.com), and connect jobseekers directly to their desired career options. As a result, students preferred to use job search websites that would cater to specific university majors, were equipped with bettercustomised search options, and were more directly linked with specific companies' vacancies. For example, although a job search website that addresses jobs across disciplines (e.g., Monster.com) is easy to use and has a customized search engine, it does not link jobseekers with specific job titles at specific companies; rather, it provides wider search results. For this reason, some students preferred industry-specific search engines such as Jobsinoilandgas.com. The findings of this study on using job search websites conform to Gunz and Peiperl's (2007) findings that students in professional degree programs did not prefer general job search websites due to their poor accuracy of results. On the other hand, although Lin (2010) found that ease of use and usefulness influenced students' choices of using job search websites, one should distinguish between how convenience and usefulness apply to online job search banks targeting general versus specific disciplines and industries. It seems that participants in this study were

able to make the distinction, but only in having prior search objectives, trials, and errors of using search strategies. Online job search websites similar to LinkedIn reduce time and increase accuracy in obtaining desired results, while general online search banks produce a massive number of inapplicable job alerts.

Student jobseekers in this study have been able to effectively target their search objectives and discover the tools that lead them to their objectives. Among these search strategies is obtaining membership to professional associations, such as Professional Engineers and Geoscientists of Newfoundland and Labrador (PEGNL) or Society of Petroleum Engineers (SPE). Once a professional association was located, they would attempt to find out how the associations could benefit them by uncovering information related to their job search; this could be in the form of receiving job alerts, learning about their networking events and services (e.g., mentorship programs), and learning about the industry. Use of this strategy demonstrates an awareness of search objectives among students; however, experienced students more actively participated in professional associations and had a better understanding of their benefits; they used them to receive emails about jobs and networking events. Students without prior experience tended to be more passive in terms of job search and relied on their networks of friends, university networking sessions, and online job searching. However, they seemed to seek the best practices of job searching by inquiry, networking, and sending direct applications.

Of the various job search strategies discussed so far, the choice of strategy was influenced by whether the students had prior work experience or not. It seems that previous exposure to a workplace environment enabled student jobseekers to have a better grasp of what they might experience during the job search process, simply because they had been through the process before. Therefore, students who were taking a break from work to pursue higher

education had more experience in this endeavor, which led to better job search clarity and affected their decisions to use LinkedIn and professional associations. Although identifying successful search strategies was not part of the research plan, data from experienced students indicated references to successful and effective job search strategies. Thus, those early in their career search would benefit from consulting people with previous experience on successful search strategies.

Students without prior experience demonstrate exploratory behaviour when it comes to search strategies. Although an exploratory search is a more diverse career cruising (Thomas, Crossley & Highhouse, 2005), the nature of this strategy is to explore different options, whether career- or search-related, because of a broader range of objectives due to lack of experience. Conversely, experienced students used a more focused search strategy (Crossley & Highhous, 2005), since they had clear career and search objectives.

Bissola and Imperatori's study on job search strategies of generation Y members (2013) focused on participants aged 25–30. They found that generation Yers used company websites, the university career centre (UCC), and LinkedIn the most. Other studies on job search strategies used by accounting graduates (Kimmell, et al., 2003) and MBA graduates (Rate & Boost, 2012) found that the UCC, business contacts, job fairs, temporary agencies, general- and major-specific job search websites, newspaper advertisements, and company websites were used the most by the students.

The importance of the UCC is highlighted in these studies and students used it as their first point of contact to look for employment. The findings of this study are validated by the aforementioned studies, considering that the participants in this research project were generation Yers and were majoring in professional programs that could have similar implications to

accounting and MBA graduates. However, this study did not report any results concerning the use of newspaper advertisements. This might be because newspaper ads are not popular anymore, are costlier to publish, and the oil and gas industry's job openings cover a wide range of positions that cannot be reduced to a few listings in newspapers.

All students were aware of the university's career resources to some extent, and most of them availed of services such as resume/cover letter writing workshops and attended various networking sessions. These findings are in contrast to the study by Gerard (2011), who found that the student participants were not aware of the UCC's services except for the resume and interview workshops. This data shows that students attempted to gain awareness of their local labour market and effective search strategies by accessing resources that connect them with job openings directly and facilitate their career search preparation. Students recognize that identifying such services at the career centre could lead them to knowledgeable individuals or to online search options that link them to companies directly. The UCC has also played a significant role in raising students' awareness of tools and resources that would benefit their search objectives. What is most relevant here is using the UCC as a search strategy, while it also acts as an influential factor in students' choices to use job search strategies by providing such strategies in significant detail. The UCC as both a job search strategy and an influence has great implications to this study. This is because the UCC presents itself as a valuable source of information on the local labour market that could connect students to industry professionals.

The UCC provides various networking opportunities for the students such as career fairs, employer information sessions, and speed networking events. Career fairs, for example, carry more information than company websites (Cable & Yu, 2006) because of the face-to-face nature of information exchange (Griffeth, Tenbrink, Robinson, 2013). In addition, the UCC's resume

and cover letter workshops are a great influence on the students' overall job search process, as they are used and favored by all the students. This is similar to the findings of Gerard (2011). As a result of their positive experiences with such services, students shared their success stories with their friends. For instance, one of the students mentioned that he was able to find a part-time job during a networking event held by the UCC. Overall, according to the UCC, international students comprise the largest group of users of on-campus career services.

The career options of student jobseekers in oil and gas are not limited to industry jobs. Some might pursue higher education to gain teaching and research positions in universities, while others might look for research positions in companies. One example is the doctoral student with industry experience who had decided to pursue higher education in order to acquire research and teaching experience. His search objectives included seeking means to enhance his skills for his future career. He attended a teaching program for graduate students (TSEP) to give him some exposure to teaching skills and the chance to build a teaching dossier as a supporting document for future employment.

Another example is the doctoral student who was interested in a research and development position (R&D) with a large oil and gas company because not all oil and gas companies are equipped with an R&D department. In the first example, in order to access vacant teaching positions, the candidate would look for online job searches that offer only teaching and research or postdoctoral positions, such as Academia.edu or Researchgate.com, versus students seeking industry employment who prefer industry-specific job search banks such as Rigzone.com and Jobsinoilandgas.com. In the second example, the student used the internet and her circle of friends to identify companies that have R&D job vacancies. In both cases, the type of job students sought affected their specific search strategies. In addition, the student interested

in teaching positions preferred to attend conferences over other networking events because of greater exposure to more academic researchers, whereas other students preferred employer information sessions and industry networking events offered by professional associations.

Oil and gas is a global industry and a list that includes most companies is available on the internet. Besides, several very large and popular oil and gas companies are present at local job fairs and information sessions and actively refer students to their websites. Therefore, students use company websites and apply directly through them because the nature of the industry makes it a visible job market. A report by Ernst and Young (2011) on HR practices in the oil and gas industry shows that advertising on company websites accounts for nearly 84% of all recruitment practices. Another instance is a master's student in geology whose research project mainly focused on offshore Newfoundland, and some of the oil and gas companies in the Province sponsored him. Although the nature of his research project does not guarantee him a position with a sponsor employer, his knowledge and expertise now lies within Canadian oil and gas industry characteristics. This student actively networked, both in person and on LinkedIn, with industry individuals within the Province. Hence, the type of industry could be another factor that influences the jobseeker's use of specific job search strategies. Industry- and job-specific search options can apply to any discipline or position; thus, it is more important that candidates in a similar situation envision clear career objectives in order to have job search clarity. They might also need to increase their networking efforts in order to locate targeted individuals with access to industry-related information. For example, the candidate with the goal to work in oil and gas R&D might exert extra effort to reach information-rich individuals because of the nature of the industry.

The literature includes several studies that describe different influencing factors in job search strategies: those that the researcher refers to as primary factors (viz., race, age, gender, education and social status); and secondary factors, which were presented without shedding light on them as influencing factors, such as information-richness, cost, convenience and usefulness, and personality traits. Secondary factors were identified in quantitative studies; in contrast, this research project provides an in-depth qualitative analysis, conducted to obtain greater insight into the participants' experiences and perceptions regarding their choices in the job search process. Finally, the findings of this study on the influencing factors are of great importance to the existing literature and carry value for student jobseekers and the UCC.

Overall, the informal job search strategy of approaching peers and friends (Cahuc & Fontaine, 2009), in addition to the UCC, can both be considered job search strategies and influencing factors. The results of this study were not intended to provide a consensus or contrast to the literature on influencing factors because of its qualitative nature. Instead, it contributed to the existing literature by providing insight into how students as potential jobseekers choose their job search strategies.

5.3 Students' Perceptions of the Oil and Gas Labour Market of the Province

The provincial government's job growth forecast of the oil and gas industry has addressed the need for a skilled workforce in the next decade (Department of Human Resources of the NL Government, 2011). Petroleum Human Resources of Canada (2013) predicted that there are three prominent factors that will lead to a great demand in the oil and gas workforce recruitment in the coming decade, i.e., "age-related attrition, workforce competition and hiring due to industry activity levels" (p. 4). The report also states that "over the next decade, the sector will need to fill between 37,700 and 47,900 job openings due to industry activity and age-related

attrition." This data has great implications for this study because the Province's oil and gas industry will undergo a shortage of skilled workers that could affect its sustainability (Petroleum Human Resources Council of Canada, 2013). This report also lays out recommendations such as hiring temporary foreign workers to overcome workforce competition. On one hand, according to international students' experiences, the oil and gas industry of the Province has a strict hiring practice regarding the residency status of the applicants; students believe that employers only hire candidates with permanent residence or citizenship statuses. On the other hand, the Atlantic Accord Implementation Acts (1986), which is an agreement between the government of Newfoundland and the government of Canada, mandate hiring local Newfoundlanders. In addition, the report refers to sourcing "skilled workers." Since international students with prior industry experience could also be considered potential skilled workforce, there is a contradiction: Why would the industry need to source its shortage of workforce by hiring temporary foreign workers from outside of Canada? We could also pose this question: How do the Petroleum Human Resources Council of Canada and the Government of Newfoundland and Labrador liaise in order to gain better access to the already existing skilled and ready-to-integrate workforce in the Province? International students spend a minimum of two years in the Province and are trying their best to integrate into society, learn about the Canadian and local labour market and employers, and enhance their skills within a short period of time while attending university. Most of them come to study in the Province with prior industry experience, yet task forces report hiring temporary foreign workers while the skilled workers already exist in the Province.

There are contradictions that place barriers in front of international students who have the desire to stay in the Province. On one hand, the Atlantic Accord Implementation Act mandate the oil and gas companies to hire local residents; on the other hand, labour market reports and task

forces, such as the Newfoundland and Labrador Skills Task Force, give recommendations on immigrants' retention in the Province, which are not consistent with the industry's hiring mandates. This being said, international students felt confident about their abilities to compete in the Canadian job market. They have an objective before starting their search process and believe they are even preparing themselves for the real and active job search upon graduation. However, it is greatly important to note their attitudes towards the current labour market. All the students showed interest to stay in Newfoundland, contingent upon finding employment in the Province; with oil and gas being a global industry with multinational offices around the world, students were willing to relocate where opportunities are available.

There were multiple student opinions addressing the lack of diversity in the hiring practices of the local oil and gas industry. "When a firm says it must increase its workforce diversity, it usually means hiring more women and ethnic minorities, or citizens of different countries" (Wilson & Iles, 1999, p. 646); however, for students, the multicultural image of a company seems to denote its employment diversity. "Diversity is important in all firms, but takes on particular importance in multinationals [because] a diverse workforce has the potential to achieve higher performance than a homogenous workforce" (Wilson & Iles, 1999, pp. 646–647). Students' perceptions on employment diversity were based on word of mouth experiences of friends and peers, suggesting that employers in the Province are not open to recruiting international students. They further reflected on this issue and reported that there are few foreign nationals working in the companies in the Province and that employers are not open to diversity. Such perceptions of the oil and gas companies may be related to the hiring mandates of the Atlantic Accord Implementation Act. As a result, international students believe they may have a better chance in other oil-hub provinces of Canada, such as Alberta. On the other hand, they have

learned that some oil and gas companies do not hire foreign nationals without permanent residence or Canadian citizenship statuses. Overall, they mentioned that they would like to remain in the Province but they are uncertain about their employment outcomes after graduation.

On another note, students' references to their top-choice oil and gas companies included multinational and large firms such as Shell, BP, and Chevron. Such multinational companies operate around the world and offer their services and products to diverse groups of consumers; however, it may be the case that their recruitment policies are shaped by their agreements with the local governments, such as the Atlantic Accord, which are reflected in the industry's policies locally. As a result, the local industry is based on the political agenda of the region shaping the companies' recruitment policies. International students consider these companies as multinational and multicultural, yet they perceive them as "hard to enter" due to their strict hiring practices surrounding the applicant's residency status.

Students' evaluation of Canadian and local hiring practices extend to the referral system. Some employers offer incentives to employees for referring jobseekers to their companies. This referral program was implemented so that companies can connect with the right talent directly. Just as jobseekers look for ways to contact employers directly to inquire about jobs, or prefer to apply to jobs directly on the company website, employers are also interested in accessing talented individuals directly to minimize hiring costs and time. Having knowledge of the referral system, students' most effective job search strategies included networking, friends, and social contacts. Although "networking does indeed lead individuals into new jobs" (Wanberg et al., 2000, p. 501), its use is recommended alongside other search strategies (Kim, 1994). The main criticism of the referral system from the students' points of view is that, if this system is practiced solely, companies cannot access a wider range of talent. However, the information-

richness associated with using networking and referrals could be a reason for jobseekers and recruiters to use these methods to "have access to particularly reliable and rich information about job characteristics ... and job applicants [respectively]" (Kim, 1994, p. 99).

Overall, our knowledge of the attitudes of the participants of this study towards the Canadian labour market is important in providing a contextual understanding of the findings. Hence, this study is a reflection on the Canadian oil and gas job market assessed through the lens of international graduate engineering and earth sciences students.

Chapter 6

Recommendations and Conclusions

This chapter presents recommendations from the researcher addressing students, the career centre, and labour market policy makers. Conclusions are given at the end of the chapter.

6.1 Recommendations from the Researcher

Based on an understanding of international students' perceptions of the local labour market and their job search behaviours, this study has implications for the labour market recruitment policy makers, career support providers at post-secondary institutes, and students. Students interviewed in this study were willing to stay in the Province of Newfoundland and Labrador upon graduation, contingent on finding employment. As a result, policymakers need to be aware of the existing workforce in the Province and match their studies and reports with the reality of its existence. On the other hand, the university career centre, as a significant resource for students' career preparation and guidance, must offer a broader branding and marketing of their services in order to reach a wider range of students. They will also need to offer more targeted, regular programs to address graduate students from a variety of disciplines.

Furthermore, offering networking sessions will be more meaningful once the career centre staff have an awareness of students' objectives and expectations in attending the sessions. It is also imperative to know how students define networking in the context of seeking employment and recognize their reasons for considering it as the most favoured job search strategy. In addition, realizing that networking is a process rather than a method in the job search process would be beneficial for career preparation timeliness and goal setting for students.

Career centre staff ensure they hear students' opinions on their services by administering a survey after each visit; however, such polling could be done prior to program design with

access to as many students as possible. Besides, reaching out to networks and groups of friends could be a successful marketing strategy for career resources. Also, students need to know that networking is more effective coupled with other search strategies, and what makes networking successful is the information exchange. Finally, partnership for career preparation purposes between the UCC and the specific departments is recommended, for example, by simply using departmental email services and planning career events in collaboration with various departments.

Another recommendation by the Newfoundland and Labrador Skills Task Force (2007) for post-secondary institutes is to "promote greater networking among industry, faculty and students" (p. 58). Networking takes place between students and employers through various networking sessions held by the UCC; however, students mentioned that they would like to see a higher occurrence of these events. On the other hand, collaboration with the faculty could be beneficial to both gain and give knowledge of the ways to integrate experiential learning with the curriculum. Also, the faculty members could be persons of contact with potential employers.

Overall, future studies can address how international students build strong and weak ties and for what purposes. Moreover, the job search strategies of students in other disciplines could add to our knowledge of post-secondary students' job search behaviours in the Province.

6.2 Conclusion

This study was conducted with the aim of addressing job search strategies of international graduate students in oil and gas engineering and earth sciences programs, factors that influenced their choices of using such strategies, and the role of higher education in their search process by locating students' uses of their services.

International graduate students in this study had objectives before starting their search process. Their purposeful search began with an intention to gain awareness about the local job market. For example, they started to build their network of human capital early in their programs and planned to connect with their list of contacts in their final semester, when they are nearing graduation, to ask if there are any openings or if the contacts are aware of any openings. Students both with and without experience were not only aware of different factors in their job search (i.e., time and preparedness) and had objectives, but they were also proactive and took thoughtful actions towards reaching their objectives while taking these factors into consideration. Their conception of job searching before graduation involved a preparatory stage, which required specification of time and identification of objectives ahead of graduation in order to prepare themselves for the active job search upon graduation.

The most noteworthy finding of this study is that students choose strategies that allow them to expose themselves to industry: talking face-to-face, approaching the people from the industry directly, or gaining valuable insight about the culture, expectations, and projects of the companies. This could extend to attending networking events offered by the UCC or professional associations, building networks of weak and strong ties, and accessing company and industry information on social media via LinkedIn. Students also used major- and field-specific job search websites, job alerts from professional associations, and they applied directly to company websites. The most influential factor on the students' job search strategies was peers and friends, followed by the UCC as a provider of various networking opportunities and other career-related services. Students defined "networking" as the opportunities to expose themselves and their skills to the companies by connecting with them either in person or through the internet. Such

opportunities could be any encounter with people, either weak or strong ties, or using LinkedIn as a medium or tool with the purpose of conveying job-related information.

Other factors that influenced students' choices of search strategy were as follows: the type of job they sought based on their research specialization or whether they were seeking academic or industry jobs; friends sharing their experiences about using successful search strategies, such as networking; the type of industry (i.e., oil and gas, which is a visible and easily accessible job market through online job banks); professional associations; student jobseekers having previous experience, which would direct them towards attending networking events and seeking membership at professional associations, versus those without prior experience who used university networking sessions; online job banks; and the influence of the career centre by providing networking opportunities and career guidance.

Overall, students were confident and purposeful in their job search, despite their perceptions of lack of diversity and unequal hiring practices regarding residency status. They also considered that building knowledge of the local labour market was beneficial for their active job search after graduation, and they practiced this by face-to-face and online networking.

References

- Adams, S. (2013). New Survey: LinkedIn More Dominant than Ever Among Job Seekers and Recruiters, but Facebook Poised To Gain. Forbes, 10, 28–29. Retrieved from <u>http://www.forbes.com/sites/susanadams/2013/02/05/new-survey-linked-in-more-</u> dominant-than-ever-among-job-seekers-and-recruiters-but-facebook-poised-to-gain/
- Archambault, A., & Grudin, J. (2012). A longitudinal study of Facebook, LinkedIn, & twitter use. In Proceedings of the SIGCHI Conference on Human Factors in Computing Systems (pp. 2741-2750). ACM.
- Barthorpe, S., & Hall, M. (2000). A collaborative approach to placement preparation and career planning for university students: a case study. *Journal of Vocational Education and Training*, 52(2), 165-175. doi: 10.1080/13636820000200113.
- Ben-Porath, Y. (1980). The F-connection: families, friends and firms, and the organization of exchange. Population and Development Review 6, 1–9.
- Berg, B. L., & Lune, H. (2004). *Qualitative research methods for the social sciences*. (Vol. 5).Boston: Pearson.
- Bissola, R., & Imperatori, B., (2013). Recruiting Gen Yers through Social Media: Insights from The Italian Labour Market. In Social media in human resources management. Emerald, ISBN 978-1-78190-900-3. p. 59-81.
- Boswell, W. R., Boudreau, J. W., & Dunford, B. B. (2004). The outcomes and correlates of job search objectives: searching to leave or searching for leverage? *Journal of Applied Psychology*, 89(6), 1083. doi: 10.1037/0021-9010.89.6.1083.

Boswell, W. R., Zimmerman, R. D., & Swider, B. W. (2012). Employee job search toward an understanding of search context and search objectives. *Journal of Management*, *38*(1), 129-163.

Bradshaw, T. F. (1973). Jobseeking methods used by unemployed workers. *Monthly Lab. Rev.*, 96, 35. Retrieved from <u>http://search.ebscohost.com/login.aspx?direct=true&AuthType=ip,url,uid&db=bth&AN=</u> 5990286&site=ehost-live&scope=site

- Cable, D. M., & Yu, K. Y. T. (2006). Managing job seekers' organizational image beliefs: The role of media richness and media credibility. *Journal of Applied Psychology*, 91(4), 828. doi: 10.1037/0021-9010.91.4.828.
- Cahuc, P., & Fontaine, F. (2009). On the efficiency of job search with social networks. *Journal of Public Economic Theory*, *11*(3), 411-439. doi: 10.1111/j.1467-9779.2009.01416.x.
- Canada, and Newfoundland and Labrador. *Atlantic Accord Implementation Acts*. (1986). Ottawa, On, and St. John's, NL.

Canadian Association of Petroleum Producers (CAPP). (2014). Canadian Economic

Contribution. Retrieved from

http://www.capp.ca/energySupply/nationalEconomicContribution/Pages/default.aspx

- Chua, V. (2007). The conditional effects of social networks on job earnings in highly meritocratic Singapore. *Conference Papers -- American Sociological Association*, 1. Retrieved from <u>http://citation.allacademic.com/meta/p182448_index.html</u>
- Creswell, J. W. (2012). *Educational research: Planning, conducting, and evaluating quantitative* (4th ed.). University of Nebraska-Lincoln: Pearson.

Cross, P. (2007). Recent trends in output and employment. Statistics Canada.

- Crossley, C. D., & Highhouse, S. (2005). Relation of job search and choice process with subsequent satisfaction. *Journal of Economic Psychology*, 26(2), 255-268.
- Department of Advanced Education and Skills. (2007). *All the Skills to Succeed: Report of the Newfoundland and Labrador Skills Task Force*. Newfoundland and Labrador. Retrieved from <u>http://www.aes.gov.nl.ca/publications/postsecondary/stfreport.pdf</u>
- Department of Finance, Newfoundland & Labrador Statistics Agency. (March 2014). Industry Statistics. (NAICS) Code and Employment Size Newfoundland and Labrador. Retrieved from <u>http://www.stats.gov.nl.ca/Statistics/Trade/PDF/BR_NL_NAICS_Size_2013.pdf</u>
- Department of Finance, Newfoundland & Labrador Statistics Agency. (March 2014). Industry Statistics. (NAICS) Code and Employment Size St. John's Census. Retrieved from <u>http://www.stats.gov.nl.ca/Statistics/Trade/PDF/BR_CMA_NAICS_Size_2013.pdf</u>
- Department of Finance, Newfoundland & Labrador Statistics Agency. (2015). Labour Market Statistics. Retrieved from

http://www.stats.gov.nl.ca/statistics/Labour/PDF/Employment_Industry_Monthly.pdf

Department of Human Resources, Labour and Employment. (2011). *Newfoundland and Labrador Labour Market OUTLOOK 2020.* St. John's, NL. Retrieved from <u>http://www.aes.gov.nl.ca/publications/lmoutlook2020.pdf</u>

- Devine, T. J., & Kiefer, N. M. (1993). The empirical status of job search theory. *Labour Economics*, 1(1), 3-24. doi: 10.1016/0927-5371(95)80012-M.
- Dey, F., & Cruzvergara, C. Y. (2014). Evolution of Career Services in Higher Education. *New Directions for Student Services*, 2014(148), 5-18. doi: 10.1002/ss.20105.
- Dilley, P. (2004). Interviews and the philosophy of qualitative research. The Journal of Higher Education, 75(1), 127-132. Retrieved from <u>http://www.jstor.org/stable/3838692</u>

- Ellis, R. A., & Taylor, M. S. (1983). Role of self-esteem within the job search process. *Journal of Applied Psychology*, 68(4), 632-640. doi:10.1037/0021-9010.68.4.632.
- Ernst & Young. (2011). Human resources in Canada's oil and gas sector: A snapshot of challenges and directions. Retrieved from <u>www.ey.com/ca</u>
- Feldman, D. C., & Klaas, B. S. (2002). Internet job hunting: A field study of applicant experiences with on-line recruiting. *Human resource management*, *41*(2), 175-192.
- Ford, D. R., & Purdue Univ., L. n. (1974). Job Search in the Labour Market for College Graduates: A Post Hoc Analysis and Some New Insights. Paper No. 483. Retrieved from <u>http://files.eric.ed.gov/fulltext/ED106269.pdf</u>
- Franzen, A., & Hangartner, D. (2006). Social networks and labour market outcomes: The nonmonetary benefits of social capital. European Sociological Review, 22(4), 353-368. doi:10.1093/esr/jcl001.
- Garge, R., & Telang, R. (2011). To be or not to be linked on LinkedIn: Online social networks and job search. *Social Science Research Network*.

doi: http://dx.doi.org/10.2139/ssrn.1813532

- Gatewood, R. D., Gowan, M. A., & Lautenschlager, G. J. (1993). Corporate image, recruitment image and initial job choice decisions. *Academy of Management journal*, *36*(2), 414-427. doi: 10.2307/256530.
- Gerard, J. G. (2011). Linking in with LinkedIn®: Three exercises that enhance professional social networking and career building. *Journal of Management Education*, doi: 10.1177/1052562911413464.
- Granovetter, M. (1973). *The strength of weak ties*. American Journal of Sociology, 78, 1360-1380.

- Granovetter, M. (1983). The strength of weak ties: A network theory revisited. *Sociological Theory*, 1, 201-233. Retrieved from <u>http://www.jstor.org/stable/202051</u>
- Granovetter, M. (1995). *Getting a job: A study of contacts and careers*. University of Chicago Press.
- Grenon, L. (1999). Obtaining a job. *Perspectives on Labour and Income*, *11*(1), 23-27. Retrieved from http://search.proquest.com/docview/213994606?accountid=12378

Griffeth, R. W., Tenbrink, A., Robinson, S. (2013). Recruitment sources: A review of outcomes.In Cable, D. M. (Ed.). (2013). *The Oxford Handbook of Recruitment* (pp. 215-250).Oxford University Press.

- Groenewald, T. (2004). A phenomenological research design illustrated. University of Alberta. Retrieved from <u>http://hdl.handle.net/10500/2573</u>
- Guerrero, L., & Rothstein, M. G. (2012). Antecedents of underemployment: Job search of skilled immigrants in Canada. *Applied Psychology*, 61(2), 323-346. doi: 10.1111/j.1464-0597.2011.00470.x.

Gunz, H. P., & Peiperl, M. (2007). Handbook of career studies. Sage Publications.

Hempel, J. (2010). How LinkedIn will fire up your career. *FORTUNE Magazine*. 161, 7: 74. Retrieved from

http://archive.fortune.com/2010/03/24/technology/linkedin_social_networking.fortune/in dex.htm

Herbold, J., & Douma, B. (2013). Students' Use of Social Media for Job Seeking. CPA Journal, 83(4), 68-71. Retrieved from <u>http://search.ebscohost.com/login.aspx?direct=true&AuthType=ip,url,uid&db=bth&AN= 87368460&site=ehost-live&scope=site</u>

- Hill, R. H. (1983). The meaning of work and the reality of unemployment in the Newfoundland context. Community Services Council Newfoundland and Labrador.
- Holzer, H. J. (1986). Search method use by unemployed youth. *Journal of Labour Economics*.Vol. 6, No. 1. pp. 1-20. Retrieved from http://www.jstor.org/stable/2534865
- Huffman, M. L., & Torres, L. (2001). Job search methods: Consequences for gender-based earnings inequality. *Journal of Vocational Behaviour*, 58(1), 127-141. doi: 10.1006/jvbe.2000.1770.
- Hycner, R. H. (1985). Some guidelines for the phenomenological analysis of interview data. *Human studies*, 8(3), 279-303. Retrieved from <u>http://www.jstor.org/stable/20008948</u>
- Johnson, J. (1997). Chasing the dream: Job search strategies, employment outcomes, and employer perceptions of recent college graduates (Order No. 9804221). Available from ProQuest Dissertations & Theses Full Text. (304388032). Retrieved from <u>http://search.proquest.com/docview/304388032?accountid=12378</u>
- Karlin, S. (2005). Networking know-how online networking. *IEEE Spectrum*, 42(11) doi: http://dx.doi.org/10.1109/MSPEC.2005.1526908
- Kibbe, C. (2009). Social networking media enhances your job search.(JOB SEARCH). New Hampshire Business Review, 31(24), 12. Retrieved from <u>http://search.ebscohost.com/login.aspx?direct=true&AuthType=ip,url,uid&db=bwh&AN</u> <u>=45059883&site=ehost-live&scope=site</u>
- Kim, J. (1994). Job search methods: Use and effectiveness. (Electronic Doctoral Dissertation). Retrieved from <u>https://etd.ohiolink.edu/</u> (Accession Order No. [osu1265296515])

- Kimmell, S. L., Keltyka, P. K., Olsen, D. H., & Kaplan, B. (2003). Accounting graduates' job search strategies. *CPA Journal*, 73(4), 72. Retrieved from <u>http://search.ebscohost.com/login.aspx?direct=true&AuthType=ip,url,uid&db=bth&AN=</u> 9529397&site=ehost-live&scope=site
- Koen, J., Klehe, U., & Van Vianen, A. M. (2012). Training Career Adaptability to Facilitate a Successful School-to-Work Transition. *Journal of Vocational Behaviour*, 81(3), 395-408. doi: 10.1016/j.jvb.2012.10.003.
- Kuhn, P., & Skuterud, M. (2002). Internet job search and unemployment durations. Retried from http://hdl.handle.net/10419/21214
- Larkin, M., Watts, S., & Clifton, E. (2006). Giving voice and making sense in interpretative phenomenological analysis. *Qualitative research in psychology*,3(2), 102-120. doi: 10.1191/1478088706qp062oa
- Legard, R., Keegan, J., & Ward, K. (2003). In-depth interviews. Qualitative research practice: A guide for social science students and researchers, 138-169. Retrieved from http://www.scope.edu/Portals/0/progs/med/precoursereadings/IEIKeyReading5.pdf
- Lester, S. (1999). An introduction to phenomenological research. Stan Lester Developments, 1-4. Retrieved from <u>www.sld.demon.co.uk/resmethy.pdf</u>
- Lin, H. F. (2010). Applicability of the Extended Theory of Planned Behaviour in Predicting Job Seeker Intentions to Use Job-Search Websites. *International Journal of Selection and Assessment*, 18(1), 64-74. doi: 10.1111/j.1468-2389.2010.00489.x.
- Lin, N., & Dumin, M. (1986). Access to occupations through social ties. *Social networks*, 8(4), 365-385. doi: 10.1016/0378-8733(86)90003-1.

Lockard, C. B., & Wolf, M. (2012). Occupational employment projections to 2020. Monthly Lab. Rev., 135, 84. Retrieved from <u>http://web.a.ebscohost.com/ehost/pdfviewer/pdfviewer?sid=9f6849d4-4f2f-4508-a2b0-a8b1ad7087da%40sessionmgr4003&vid=2&hid=4206</u>

- Lynch, G. (n.d). Sampling. University of Kent. Retrieved from http://www.kent.ac.uk/religionmethods/documents/Sampling.pdf
- Marmaros, D., & Sacerdote, B. (2002). Peer and social networks in job search. *European* economic review, 46(4), 870-879. doi: 10.1016/S0014-2921(01)00221-5.
- McKenna, C. J. (1985). Uncertainty and the labour market: Recent developments in job-search theory. Brighton: Wheatsheaf.
- McKeown, T., & Lindorff, M. (2011). The graduate job search process-a lesson in persistence rather than good career management?. *Education+ Training*,53(4), 310-320. doi: 10.1108/00400911111138479.
- McIlveen, P., & Pensiero, D. (2008). Transition of graduates from backpack-to-briefcase: A case study. *Education & Training*, 50(6), 489-499.

doi: <u>http://dx.doi.org/10.1108/00400910810901818</u>.

Memorial University of Newfoundland - Fact Book 2014, Table 5A, Undergraduate and Graduate Enrolment by Faculty/School - Fall Semester. Retrieved from,

http://www.mun.ca/ciap/Analysis/Factbook/Factbook2014.php

Memorial University of Newfoundland - Fact Book 2014, Table 13, Out-of-province Full-time and Part-time Undergraduate and Graduate Enrolment - Fall Semester. Retrieved from, <u>http://www.mun.ca/ciap/Analysis/Factbook/Factbook2014.php</u>

- Monahan, M., Shah, A., & Jochum, R. (2014). Job hunting strategies in the lion city. *Journal of International Business and Cultural Studies*, 8, 1-9. Retrieved from http://search.proquest.com/docview/1545106575?accountid=12378.
- Morgan, G., & Smircich, L. (1980). The case for qualitative research. Academy of management review, 5(4), 491-500. doi: 10.5465/AMR.1980.4288947.
- Nasser, R. N., & Abouchedid, K. (2006). Job-seeking behaviour and job outcomes among
 Lebanese university graduates in private and public universities. *International Journal for Educational and Vocational Guidance*,6(3), 167-180. doi: 10.1007/s10775-006-9104-2
- Newfoundland and Labrador Oil and Gas Industries Association (NOIA). (n.d). *Projects*. Retrieved from <u>http://www.noia.ca/Industry-Info/Projects/</u>
- Nikolaou, I. (2014). Social Networking Web Sites in Job Search and Employee Recruitment. *International Journal of Selection and Assessment*, 22(2), 179-189. doi: 10.1111/ijsa.12067.
- Ng, E. S., & Burke, R. J. (2006). The next generation at work-business students' views, values and job search strategy: implications for universities and employers. *Education+ Training*, 48(7), 478-492. doi: 10.1108/00400910610705872.
- O'Connor, B. N. (2004). The workplace learning cycle: a problem-based curriculum model for the preparation of workplace learning professionals. *Journal of Workplace Learning*, 16(6), 341-349. doi: 10.1108/13665620410550312.
- OECD (1999). Thematic Review of the Transition from Initial Education to Working Life: Canada. Retrieved from <u>http://www.oecd.org</u>

- Oil & Gas Development Partnership. (2013). Overview. Retrieved in July 2014 from http://www.mun.ca/OGDP/about
- Osberg, L. (1993). Fishing in different pools: job-search strategies and job-finding success in Canada in the early 1980s. *Journal of labour economics*, 348-386. Retried from http://www.jstor.org/stable/2535285
- Patton, M. (1990). Designing qualitative studies. In *Qualitative evaluation and research methods* (pp. 169-186). Beverly Hills, CA: Sage.

Petro Canada. (2007). Annual Information Form. Retrieved from

http://www.suncor.com/pdf/2006-AIF-Final-E.pdf

- Petroleum Human Resources Council of Canada. (2013). *The Decade Ahead: Labour Market Outlook to 2022 for Canada's Oil and Gas Industry*. Retrieved from <u>www.petrohrsc.ca</u>
- Petroleum Industry Human Resources Committee. (February, 2001). Analysis of Gaps and Issues Related to Labour Supply and Demand in Offshore Exploration and Production in Newfoundland. Retrieved from <u>http://www.oilandgascareerinfo.ca/wp-</u> content/uploads/2011/06/SkillsGapAnalysis.pdf
- Rate, J. P., & Boost, S. (2012). Alumni Job Search Strategies, Class of 2011. GMAC[R] Data-to-Go Series. Graduate Management Admission Council.
- Royal, M. (2001). Methods of job search: How university students search for part-time jobs. (Unpublished master's thesis). Saint Mary's University, Canada.
- Sanyal, R. N., & Neves, J. S. (1998). Networking: A simulation of job search behaviour. Simulation & Gaming, 29(2), 260-264. doi:10.1177/1046878198292009.
- Scheele, K., & Madison, R. (2010). Navigating the 21st century job search. Montvale: Institute of Management Accountants.

- Seyfrit, C. L. (1993). Hibernia's Generation: Social Impacts of Oil Development on Adolescents in Newfoundland. Institute of Social and Economic Research, Memorial University of Newfoundland.
- Shantz, A., Wright, K., & Latham, G. (2011). Networking with boundary spanners: a quasi-case study on why women are less likely to be offered an engineering role. *Equality, Diversity* and Inclusion: An International Journal, 30 (3), 217-232. http://dx.doi.org/10.1108/02610151111124950
- Silliker, S. (1993). The role of social contacts in the successful job search. *Journal of Employment Counseling*, *30*(1), 25-34. doi: 10.1002/j.2161-1920.1993.tb00963.x.
- Singh, A., Devine, M. (2013). Rural transformation and Newfoundland and Labrador diaspora: Grandparents, grandparenting, community and school relations. Rotterdam: Sense Publishers.
- Sin, S. J., & Kim, K. (2013). International students' everyday life information seeking: The informational value of social networking sites. Library and Information Science Research, 35(2), 107-116. doi:10.1016/j.lisr.2012.11.006.
- Skeard, J., Holloway, C., & Vodden, K. (2011). The analysis of labour market factors impacting primary industries in the province of Newfoundland and Labrador, with a special interest in the Clarenville-Bonavista rural secretariat region. *Memorial University of Newfoundland and the Government of Newfoundland and Labrador*. Retrieved from <u>http://www.exec.gov.nl.ca</u>
- Smith, J.A. and Osborn, M. (2003) Interpretative phenomenological analysis. In J.A. Smith (Ed), Qualitative psychology: A practical guide to research methods. London: Sage. Retrieved from <u>http://dx.doi.org/10.1080/14780880802314304</u>

Statistics Canada. (2012). North American Industry Classification System (NAICS) Canada. Retrieved from

http://www23.statcan.gc.ca/imdb/p3VD.pl?Function=getVD&TVD=118464&CVD=118 467&CPV=2111&CST=01012012&CLV=3&MLV=5

Suki, N. M., Ramayah, T., & Ming, M. K. P. (2011). Job searches via social networking sites:
Employed job seekers intentions. In *Handbook of Research on Business Social Networking: Organizational, Managerial, and Technological Dimensions: Organizational, Managerial, and Technological Dimensions*, 418.

- Sułkowski, Ł., & Chmielecki, M. 2014. The use of Social Networking Sites in modern recruitment–empirical study. *Przedsiebiorczosc i Zarzadzanie*, 15(2), 63-73. Retrieved from <u>http://piz.san.edu.pl/docs/e-XV-2.pdf#page=63</u>
- Suncor Energy INC. (2012). Annual Report. Retrieved from http://www.suncor.com/pdf/Suncor_Annual_Report_2012_en.pdf
- The Globe and Mail (Canada). Smith, S. (2008). Co-op studies a 'try before you buy' opportunity. Retrieved in July 2014 from http://www.theglobeandmail.com/report-on-business/co-op-studies-a-try-before-you-buy-opportunity/article664030/
- Thomas, L. C., (1989). *Self-Esteem: Implications for Job Expectations and Job Search Strategies*. The Ohio State University.
- Today.mun.ca. (2014). \$2.3 million to overcome challenges in oil and gas sector. Retrieved from http://today.mun.ca/news.php?id=9089
- Try, S. (2005). The use of job search strategies among university graduates. *The Journal of Socio-Economics*, *34*(2), 223-243. doi: 10.1016/j.socec.2004.09.009.

Van Hoye, G. (2009). Networking as a job search behaviour: A social network perspective. *Journal of Occupational & Organizational Psychology*, 82(3), 661-682. doi: 10.1348/096317908X360675.

- Van Hoye, G. (2013). Word of mouth as a recruitment source: An integrative model. In Cable,D. M. (Ed.). (2013). *The Oxford Handbook of Recruitment* (pp. 251-268). Oxford University Press.
- Van Hoye, G., Klehe, I. U., & Van Hooft, E. A. J. (2014). Job search behaviour as a multidimensional construct: A review of different job search behaviours and sources.
 Oxford Handbook of Job Loss and Job Search (in press). New York: Oxford University Press. Retrieved from http://www.researchgate.net
- Van Hoye, G., & Saks, A. M. (2008). Job search as goal-directed behaviour: Objectives and methods. *Journal of Vocational Behaviour*, *73*(3), 358-367.
 doi: 10.1016/j.jvb.2008.07.003.
- Wanberg, C. R., Hough, L. M., & Song, Z. (2002). Predictive validity of a multidisciplinary model of reemployment success. *Journal Of Applied Psychology*, 87(6), 1100-1120. doi:10.1037/0021-9010.87.6.1100
- Wanberg, C. R., Kanfer, R., & Banas, J. T. (2000). Predictors and outcomes of networking intensity among unemployed job seekers. *Journal of Applied Psychology*, 85(4), 491. doi: 10.1037/0021-9010.85.4.491.
- Weber, A., & Mahringer, H. (2008). Choice and success of job search methods. *Empirical Economics*, 35(1), 153-178. doi: 10.1007/s00181-007-0148-z.

Wei-Cheng, M., & Kopischke, A. (2001). Job search methods, job search outcomes, and job satisfaction of college graduates: A comparison of race and sex. *Journal of Employment Counseling*, 38(3), 141-149.

doi: 10.1002/j.21611920.2001.tb00496x

- White, H. C., Boorman, S. A., & Breiger, R. L. (1976). Social structure from multiple networks.
 I. Blockmodels of roles and positions. *American journal of sociology*, 730-780. Retrieved from http://www.jstor.org/stable/2777596
- Whitt, E. J. (1991). Artful Science: A primer on qualitative research methods. *Journal of College Student Development*, 32(5), 406-15. Retrieved from http://eric.ed.gov/?id=EJ435443
- Wilden, R., Gudergan, S., & Lings, I. (2010). Employer branding: strategic implications for staff recruitment. *Journal of Marketing Management*, 26(1-2), 56-73.
 doi: 10.1080/02672570903577091.
- Wilson, E. M., & Iles, P. A. (1999). Managing diversity-an employment and service delivery challenge. *International Journal of Public Sector Management*,12(1), 27-49. doi:10.1016/S0263-2373(98)00041-3
- Yamkovenko, B., & Hatala, J. P. (2014). Feedback-seeking and social networking behaviours during job search a conceptual model. *Human Resource Development Review*, 13(1), 102-124.

Bibliography

- American Psychological Association. (2010). Publication manual of the American Psychological Association. (6th ed.). *Washington, D.C.:* American Psychological Association.
- Blau, G. (1994). Testing a two-dimensional measure of job search behaviour. Organizational behaviour and human decision processes, 59(2), 288-312.
 doi: 10.1006/obhd.1994.1061.
- Canada-Nova Scotia offshore petroleum board. (2012-2013). Annual Report: Leading by Efficient, Fair and Competent Regulation. CNSBPB: Halifax, Canada.
- Cook, T., & Rushton, B. S. (2010). *How to recruit and retain higher education students: A handbook of good practice*. Routledge.
- Dupre, R. (2014). Apprenticeships Gain Momentum in Upstream Oil, Gas Sector. Retrieved in July 2014 from

http://www.rigzone.com/news/oil_gas/a/133646/Apprenticeships_Gain_Momentum_in_ Upstream_Oil_Gas_Sector/?all=HG2#sthash.YB0sf713.dpuf

- Hamilton, L. C., & Seyfrit, C. L. (1994). Resources and hopes in Newfoundland. *Society & Natural Resources*, 7(6), 561-578. doi: 10.1080/08941929409380889.
- Howe, N., & Strauss, W. (2000). *Millennials rising: The next great generation*. New York: Vintage Books.
- Ibarra, H., & Andrews, S. B. (1993). Power, social influence, and sense making: Effects of network centrality and proximity on employee perceptions. *Administrative science quarterly*, 277-303. doi: 10.2307/2393414.
- Maki, D. R. (1972). *Search behaviour in Canadian job markets*. Special study No. 15. Ottawa: Economic Council of Canada.

- Saks, A. M. (2006). Multiple predictors and criteria of job search success. *Journal of Vocational Behaviour*, 68(3), 400-415. doi: 10.1016/j.jvb.2005.10.001.
- Southampton Solent University. (n.d.) Dissertation proposals & writing dissertations. Retrieved from <u>http://mycourse.solent.ac.uk/mod/book/view.php?id=2744&chapterid=1290</u>
- Zottoli, M. & Wanous, J. (2000). Recruitment source research: Current status and future directions. Human Resource Management Review, 10(4), 353-382. doi: 10.1016/S1053-4822(00)00032-2.
- Weller, I., Michalik, A., Mühlbauer, D. (2013). Recruitment source implications for organizational tenure. In Cable, D. M. (Ed.). *The Oxford Handbook of Recruitment* (pp. 139-160). Oxford University Press.
- Wiesenberg, F., & Aghakhani, A. (2007). An exploration of graduate students' career transition experiences. *Canadian Journal of Counselling and Psychotherapy/Revue Canadienne de Counseling et de Psychothérapie*, 41(2).

Appendices

Appendix A

Printed on Memorial University of Newfoundland letter head

Informed Consent Form-Students

Title: International Students' Job Search Strategies in the Oil and Gas Industry of Newfoundland and Labrador

Researcher: Sepideh Gorjizadeh

Faculty of Education

s.gorjizadeh@mun.ca

709-740-0335

Supervisor: Dr. Dale Kirby

Faculty of Education

dkirby@mun.ca

You are invited to take part in a research project entitled "International Students' Job Search Strategies in the Oil and Gas Industry of Newfoundland and Labrador."

This form is part of the process of informed consent. It should give you the basic idea of that the research is about and what your participation will involve. It also describes your right to withdraw from the study at any time. In order to decide whether you wish to participate in this research study, you should understand enough about its risks and benefits to be able to make an informed decision. This is the informed consent process. Take time to read this carefully and to understand the information given to you. Please contact the researcher, Sepideh Gorjizadeh, if

you have any questions about the study or for more information not included here before you consent.

It is entirely up to you to decide whether to take part in this research. If you choose not to take part in this research or if you decide to withdraw from the research once it has started, there will be no negative consequences for you, now or in the future.

Introduction:

I am Master's student in the Education Faculty. As part of my Master's thesis, I am conducting this research under the supervision of Dr. Dale Kirby.

This study is aimed at bridging the gap in our knowledge of the job search strategies used by the postgraduate students in Oil & Gas and Earth Sciences degrees of the Province by understanding the factors affecting their choices of job seeking strategies, and identifying the role of the higher institutions in the Province in their job search process.

This research will be shared in academic conferences and publications and will contribute to the students' knowledge of job search strategies, and will provide recommendations to the higher institutes' student career preparation plans as well as recruiters' recruitment strategies.

Purpose of Study:

The purpose of this study is to a) provide an understanding of student job search strategies in oil and gas industry in the Province of Newfoundland and Labrador, b) identify the factors affecting the choices that the students make in order to search for and find jobs, and c) recognize supports, guidance, and resources that higher education institutions in Newfoundland and Labrador provide to students who are seeking employment opportunities in the oil and gas industry.

What You Will Do in This Study:

You will be interviewed on a questionnaire prepared that addresses the job search strategies that you use and the factors affecting those strategies, as well as your knowledge of the career services and programs available to you on campus.

Length of Time:

The interview will take approximately 40 minutes to complete.

Withdraw from the Study:

You will voluntarily agree to be involved in this face-to-face interview process. You may end your involvement before or during the interview without any negative consequences simply by saying that you wish to stop and withdraw. With consent, the researcher will retain and use for research purposes the data collected up to the point of withdrawal. If permission is not granted, the researcher will delete and dispose of the relevant data.

After the interview, the participant may request the interview transcript to be removed from the research only within three weeks from the date of the interview before the data is aggregated for analysis. All the data for aggregation will be anonymized.

Possible Benefits:

Your participation in this study will help you identify a variety of job search strategies. You will have the opportunity to ask questions regarding job search process as well. Also, you will become familiar with student career services/programs on campus.

Possible Risks:

You should not experience any risks of being in the study. It will take only a short amount of your time at a location and time convenient to you.

Confidentiality vs. Anonymity:

This interview process is confidential and the identities of participants will be accessible only to the researchers. Responses will be structured through analysis and quotations will be attributed through pseudonyms.

Confidentiality and Storage of Data:

Your interview responses will not include your name. It will be added to the pile of interviews conducted with other student participants in this face-to-face manner. Your interview, with your consent, will be audio-recorded and transcribed by the interviewer. The recordings and transcriptions will be stored in a password-protected folder and laptop. All electronic data, including interview soundtracks and transcriptions will be retained for five years in accordance with Memorial University's policy on Integrity in Scholarly Research.

Reporting of Results:

The data collected in your interview will be used and published as my master's thesis as well as in journal articles, conference presentations, and oral reports within the Memorial University community. Also, the thesis will be publically available at the QEII library. The data you provide will not be attributed to your identity and the direct quotations will be used with pseudonyms, if permission granted, otherwise your data will be reported only in a summarized manner.

Sharing of Results with Participants/Questions:

You may obtain information or the results of this study in the future by contacting the researcher at <u>s.gorjizadeh@mun.ca</u> or 709-740-0335

ICEHR Approval Statement

The proposal for this research has been reviewed by the Interdisciplinary Committee on Ethics in Human Research and found to be in compliance with Memorial University's ethics policy. If you

have ethical concerns about the research (such as the way you have been treated or your rights as a participant), you may contact the Chairperson of the ICEHR at icehr@mun.ca or by telephone at 709-864-2861.

Consent

Your signature on this form means that:

- You have read the information about the research.
- You have been able to ask questions during this study.
- You are satisfied with the answers to all your questions.
- You understand what the study is about and what you will be doing.
- You understand that you are free to withdraw from the study at any time, without having to give a reason, and that doing so will not affect you now or in the future.
- You understand that any data collected from you up to the point of your withdrawal will be retained by the researcher for use in the research study unless you request otherwise.

If you sign this form, you will not give up your legal rights and do not release the researcher

from their professional responsibilities.

Your signature means:

I have read what this study is about Yes No and understood the risks and benefits.
I have had adequate time to think about this and had the opportunity to ask questions and my questions have been answered.

	0	I agree to participate in this study	Yes No		
		understanding the risks and			
		contributions of my participation, that			
		my participation is voluntary, and that			
		I may end my participation at any			
		time.			
	0	I agree to be audio-recorded during	Yes No		
		the interview.			
	0	I agree to the use of quotations	Yes No		
		without my name being identified in			
		any publications resulting from this			
		study.			
	0	A copy of this Consent Form has been	Yes No		
		given to me for my record.			
	Signature of participant		Date		
Researcher's signature					
	I have explained this study to the best of my ability. I invited questions and gave answers.				
I believe that the participant fully understands what is involved in being in the study, any					

potential risks of the study and that he or she has freely chosen to be in the study.

Signature of principal investigator

Date

Appendix B

Printed on Memorial University of Newfoundland letter head

Informed Consent Form-Administration

Title: International Students' Job Search Strategies in the Oil and Gas Industry of Newfoundland and Labrador

Researcher: Sepideh Gorjizadeh

Faculty of Education

s.gorjizadeh@mun.ca

709-740-0335

Supervisor: Dr. Dale Kirby

Faculty of Education

dkirby@mun.ca

You are invited to take part in a research project entitled "International Students' Job Search Strategies in the Oil and Gas Industry of Newfoundland and Labrador."

This form is part of the process of informed consent. It should give you the basic idea of that the research is about and what your participation will involve. It also describes your right to withdraw from the study at any time. In order to decide whether you wish to participate in this research study, you should understand enough about its risks and benefits to be able to make an informed decision. This is the informed consent process. Take time to read this carefully and to understand the information given to you. Please contact the researcher, Sepideh Gorjizadeh, if you have any questions about the study or for more information not included here before you consent.

It is entirely up to you to decide whether to take part in this research. If you choose not to take part in this research or if you decide to withdraw from the research once it has started, there will be no negative consequences for you, now or in the future.

Introduction:

I am Master's student in the Education Faculty. As part of my Master's thesis, I am conducting this research under the supervision of Dr. Dale Kirby.

This study is aimed at bridging the gap in our knowledge of the job search strategies used by the postgraduate students in Oil & Gas and Earth Sciences degrees of the Province by understanding the factors affecting their choices of job seeking strategies, and identifying the role of the higher institutions in the Province in their job search process.

This research will be shared in academic conferences and publications and will contribute to the students' knowledge of job search strategies, and will provide recommendations to the higher institutes' student career preparation plans as well as recruiters' recruitment strategies.

Purpose of Study:

The purpose of this study is to a) provide an understanding of student job search strategies in oil & gas industry in the Province of Newfoundland and Labrador, b) identify the factors affecting the choices that the students make in order to search for and find jobs, and c) recognize supports, guidance, and resources that higher education institutions in Newfoundland and Labrador provide to students who are seeking employment opportunities in oil and gas industry.

What You Will Do in This Study:

You will be interviewed on a questionnaire prepared that addresses the career preparation programs and services offered by the Career and Experiential Learning Centre at MUN. You will also be asked to give recommendations on the possible future career programs for the students.

Length of Time:

The interview will take approximately 40 minutes to complete.

Withdraw from the Study:

You will voluntarily agree to be involved in this face-to-face interview process. You may end your involvement before or during the interview without any negative consequences simply by saying that you wish to stop and withdraw. With consent, the researcher will retain and use for research purposes the data collected up to the point of withdrawal. If permission is not granted, the researcher will delete and dispose of the relevant data.

After the interview, the participant may request the interview transcript to be removed from the research only within three weeks from the date of the interview before the data is aggregated for analysis. All the data for aggregation will be anonymized.

Possible Benefits:

Your participation in this study will give you an opportunity to reflect on career services/programs you offer to students and eventually suggest recommendations for future career preparation projects.

Possible Risks:

You should not experience any risks of being in the study. It will take only a short amount of your time at a location and time convenient to you.

Confidentiality vs. Anonymity:

This interview process is confidential and the identities of participants will be accessible only to the researchers. Responses will be structured through analysis and quotations will be attributed through pseudonyms. However, anonymity cannot be guaranteed because informed readers may be able to identify responses due to the small size of Memorial University's career centre employees.

Confidentiality and Storage of Data:

Your interview responses will not include your name. It will be added to the pile of interviews conducted with other administrator participants in this face-to-face manner. Your interview, with your consent, will be audio-recorded and transcribed by the interviewer. The recordings and transcriptions will be stored in a password-protected folder and laptop. All electronic data, including interview soundtracks and transcriptions will be retained for five years in accordance with Memorial University's policy on Integrity in Scholarly Research.

Reporting of Results:

The data collected in your interview will be used and published as my master's thesis as well as in journal articles, conference presentations, and oral reports within the Memorial University community. Also, the thesis will be publically available at the QEII library. The data you provide will not be attributed to your identity and the direct quotations will be used with pseudonyms, if permission granted, otherwise your data will be reported only in a summarised manner.

Sharing of Results with Participants/Questions:

You may obtain information or the results of this study in the future by contacting the researcher at <u>s.gorjizadeh@mun.ca</u> or 709-740-0335

ICEHR Approval Statement

The proposal for this research has been reviewed by the Interdisciplinary Committee on Ethics in Human Research and found to be in compliance with Memorial University's ethics policy. If you have ethical concerns about the research (such as the way you have been treated or your rights as a participant), you may contact the Chairperson of the ICEHR at icehr@mun.ca or by telephone at 709-864-2861.

Consent

Your signature on this form means that:

- You have read the information about the research.
- You have been able to ask questions during this study.
- You are satisfied with the answers to all your questions.
- You understand what the study is about and what you will be doing.
- You understand that you are free to withdraw from the study at any time, without having to give a reason, and that doing so will not affect you now or in the future.
- You understand that any data collected from you up to the point of your withdrawal will be retained by the researcher for use in the research study unless you request otherwise.

If you sign this form, you will not give up your legal rights and do not release the

researcher from their professional responsibilities.

Your signature means:

- I have read what this study is about Yes No and understood the risks and benefits.
 I have had adequate time to think about this and had the opportunity to ask questions and my questions have been answered.
- I agree to participate in this study
 Yes No
 understanding the risks and

	contributions of my participation, that	
	my participation is voluntary, and that	
	I may end my participation at any	
	time.	
0	I agree to be audio-recorded during	Yes No
	the interview.	
0	I agree to the use of quotations	Yes No
	without my name being identified in	
	any publications resulting from this	
	study.	
0	A copy of this Consent Form has been	Yes No
	given to me for my record.	
Signature of participant		Date

Researcher's signature

I have explained this study to the best of my ability. I invited questions and gave answers.

I believe that the participant fully understands what is involved in being in the study, any

potential risks of the study and that he or she has freely chosen to be in the study.

Signature of principal investigator

Date

Appendix C

Recruitment script-Students

Title: Call for participants

Participants needed for a study at Memorial University

Hi,

I am a graduate student in the Education Faculty. As part of my Master's thesis, I am conducting a research project under the supervision of Dr. Dale Kirby entitled "*International Students' Job Search Strategies in the Oil and Gas Industry of Newfoundland and Labrador*". I would like to invite post-graduate students in their final year of study in the Oil & Gas and Earth Sciences related courses here at Memorial University to participate in my research.

The interview will take about 40 minutes and is confidential.

Title of the study: International Students' Job Search Strategies in the Oil and Gas Industry of Newfoundland and Labrador

This study is aimed at bridging the gap in our knowledge of the job search strategies used by the postgraduate students in Oil & Gas and Earth Sciences degrees of the Province by understanding the factors affecting their choices of job seeking strategies, and identifying the role of the higher institutions in the Province in their job search process.

If interested, please contact Sepideh at s.gorjizadeh@mun.ca 709-740-0335

Faculty of Education Supervisor: Dr. Dale Kirby <u>dkirby@mun.ca</u>

ICEHR Approval Statement

The proposal for this research has been reviewed by the Interdisciplinary Committee on Ethics in Human Research and found to be in compliance with Memorial University's ethics policy. If you have ethical concerns about the research (such as the way you have been treated or your rights as a participant), you may contact the Chairperson of the ICEHR at icehr@mun.ca or by telephone at 709-864-2861.

Appendix D

Recruitment script-Administration

Title: Call for participants

Participants needed for a study at Memorial University

Hi,

I am a graduate student in the Education Faculty. As part of my Master's thesis, I am conducting a research project under the supervision of Dr. Dale Kirby entitled "*International Students' Job Search Strategies in the Oil and Gas Industry of Newfoundland and Labrador*". I would like to invite the administration in different roles at the Career and Experiential Learning Center (CDEL) at Memorial University to participate in my research.

The interview will take about 40 minutes and is confidential.

Title of the study: International Students' Job Search Strategies in the Oil and Gas Industry of Newfoundland and Labrador

This study is aimed at bridging the gap in our knowledge of the job search strategies used by the postgraduate students in Oil and Gas and Earth Sciences degrees of the Province by understanding the factors affecting their choices of job seeking strategies, and identifying the role of the higher institutions in the Province in their job search process.

If interested, please contact Sepideh at s.gorjizadeh@mun.ca 709-740-0335

Faculty of Education Supervisor: Dr. Dale Kirby dkirby@mun.ca

ICEHR Approval Statement

The proposal for this research has been reviewed by the Interdisciplinary Committee on Ethics in Human Research and found to be in compliance with Memorial University's ethics policy. If you have ethical concerns about the research (such as the way you have been treated or your rights as a participant), you may contact the Chairperson of the ICEHR at icehr@mun.ca or by telephone at 709-864-2861.

Appendix E

Questionnaire for postgraduate students in oil & gas related programs:

- 1. Can you tell me about yourself? What program are you in? What year?
- 2. Where are you from? And how long have you been in St. John's/Memorial University?
- 3. What are your career prospects/objectives? What do you plan to do after graduation?
- 4. Do you have prior oil and gas work experience? If yes, where and how long?
- 5. When did you/do you plan to start looking for job?
- **6.** In which city are you looking for a job? And what have motivated you to choose this city?
- 7. What is your ideal company to work for? What has motivated your decision?
- 8. How do you look for your future job? (eg. Online job banks, career agencies, internships, co-ops, networking/referrals, print media, social media, company website, submitting applications/making job profiles on company website, public employment service, campus recruitment, contacting employers, etc.)
- **9.** What has/have motivated you to choose this particular job search activity to look for your future job? (eg. Friends and family, teachers, ads, active job search,
- **10.** Which of the mentioned job search activities do you think are the most effective for your type of future career? And how do you think you can get the necessary resources in order to access such search activity?
- **11.** What do you do to keep up to date on the current openings/job opportunities in your field of study? How do you access the information about such opportunities?
- **12.** How often do you visit the career center on campus and what services of the center have you used before/are you using now? Which ones do you recommend to a friend?

13. What do you think the university should do in order to provide you with the necessary information/services/programs in helping you look for a job in your related field of study?

Appendix F

Questionnaire for administrators in career related centers at MUN:

- Can you tell me about yourself? What do you do? How long have you been working here?
- **2.** How much interaction do you have with students on a daily basis? Is it face-to-face or through distance?
- 3. Who are your most incoming students to use the services at the center?
- 4. What services/programs at the center are used most often? What do these services/programs target? (eg. Skill enhancement, job search knowledge, access to information on job search strategies, career preparation, recruitment events, etc.)
- **5.** How often do you get post graduate students in Oil and Gas or Earth Sciences programs to use your services? And what services do they use mostly?
- 6. What are your specific services/programs for such students?
- 7. What have been the results of students using your services/programs?
- **8.** How do you reach out to post graduate students in Oil and Gas or Earth Sciences programs?
- 9. What is the students' attendance like in your targeted services/programs?
- **10.** How do the teachers in Oil and Gas or Earth Sciences programs play a role in helping the students look for a job or get connected to future employers?
- 11. What is your affiliation like with the university-industry partnership center at MUN?
- **12.** How much involvement does the oil and gas industry have in university career preparation services/programs? What level of students do they target and in what programs?